

Review Exploring the Potential of Sensing for Breast Cancer Detection

Nure Alam Chowdhury ¹, Lulu Wang ²,*⁰, Linxia Gu ¹, and Mehmet Kaya ¹,*⁰

- ¹ Department of Biomedical Engineering and Science, Florida Institute of Technology,
- Melbourne, FL 32901, USA; nurealam1743phy@gmail.com (N.A.C.); gul@fit.edu (L.G.)
- ² Biomedical Device Innovation Center, Shenzhen Technology University, Shenzhen 518118, China
- * Correspondence: lwang381@hotmail.com or wanglulu@sztu.edu.cn (L.W.); mkaya@fit.edu (M.K.)

Abstract: Breast cancer is a generalized global problem. Biomarkers are the active substances that have been considered as the signature of the existence and evolution of cancer. Early screening of different biomarkers associated with breast cancer can help doctors to design a treatment plan. However, each screening technique for breast cancer has some limitations. In most cases, a single technique can detect a single biomarker at a specific time. In this study, we address different types of biomarkers associated with breast cancer. This review article presents a detailed picture of different techniques and each technique's associated mechanism, sensitivity, limit of detection, and linear range for breast cancer detection at early stages. The limitations of existing approaches require researchers to modify and develop new methods to identify cancer biomarkers at early stages.

Keywords: biosensor; NMRI; microwave imaging; biomarker; breast cancer

1. Introduction

The abnormal growth of cells in the breast is responsible for breast cancer, which is more commonly observed in women than men [1–3]. Noninvasive cancer does not spread outside the tissue from the point at which it is generated [4]. Invasive cancer does not live in a single point; instead, it spreads and invades different healthy tissues [5]. If a patient experiences cancer for a second time after receiving treatment for the first time, the cancer is called recurrent [6]. When the cancerous cells are not confined in a single position and expand to the surrounding regime of the body, the cancer is defined as metastatic cancer [7]. Invasive ductal carcinoma and invasive lobular cancer are commonly observed [8].

In contrast, inflammatory breast cancer, tubular breast cancer [9], colloid breast cancer [10], and metaplastic breast cancer are less common among patients. The size and location of the cancerous cells are the basic parameters for recognizing the stage of breast cancer [11–14]. Stage zero means abnormal cells are observed in a specific tissue regime. Stage I means the tumor size is below 2 cm [12]. Stage IIA means that the tumor is living in the lymph nodes in the armpit with a length of two centimeters or smaller [13]. Stage IIB means the tumor still lives inside the armpit lymph nodes with a size of 2 to 5 cm [13]. Stage IIIA means the cancer can be larger than 5 cm, but this tumor can be found in the armpit lymph nodes. Stage IIIB means the tumor is any size, and this cancer has already spread to the chest wall or breast skin [12–14]. Stage IIIC means a cancer of any size has already spread to distant organs [12–14].

Global cancer statistics 2020 reported that female breast cancer cases (i.e., 2.6 million) can be observed more frequently than lung cancer cases (i.e., 2.2 million) [15]. In the United States, one woman among eight women has experienced breast cancer, which has been considered the most common cancer for women worldwide [16–18]. A regular X-ray mammogram is required for 65 percent of women after age 40. Up to forty percent of breast cancer cases are not found by mammogram but rather by someone themselves, and over twelve thousand women in the United States before the age of forty are diagnosed



Citation: Chowdhury, N.A.; Wang, L.; Gu, L.; Kaya, M. Exploring the Potential of Sensing for Breast Cancer Detection. *Appl. Sci.* **2023**, *13*, 9982. https://doi.org/10.3390/ app13179982

Academic Editor: Laura Cercenelli

Received: 3 August 2023 Revised: 24 August 2023 Accepted: 2 September 2023 Published: 4 September 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). with breast cancer each year. Men also experience breast cancer, and they do not undergo routine mammograms. So, all of this means that understanding the signs and symptoms of breast cancer and knowing what to look and feel for are critical in finding breast cancer early before it progresses [18]. Several characteristics and symptoms need to be considered for breast self-examination (BSE). It is wise to check these signs and symptoms by lying down or standing in front of a mirror with arms above the head [17]. A doctor should be contacted if anyone notices any of these signs or other changes unrelated to menstruation, pregnancy, hormones, or other medical reasons. A thickened breast area can be a sign of breast cancer [19]. One sign is a sunken or inverted nipple. For some, inverted nipples are a typical shape of the breast, but for those new, the tumor can pull the nipple inward, flatten it, or cause it to sink [20]. If there is a dimple or ridge dimple in your skin that is not caused by tight clothing, it can be due to the tumor pulling the skin inward. This will create a divot or a dimple-like shape, and if it does not go away or if it worsens, be sure to reach out to your doctor. Another sign is a change in the breast shape or breast size [21]. It is common for women to have different sizes or shapes of breasts from the left compared to the right, and breastfeeding and menstruation can cause further fluctuations, but if your size or shape change does not connect to these issues, then you want to follow up with your doctor. In relation to nipple changes or scabs, if you see a scab-like area that is white or red over the nipple area that is not related to anything like breastfeeding, be sure to reach out to your doctor [22]. A hard lump or bump is also a common sign of breast cancer [23]. Lumps in the breasts are very common and may be a sign of natural breast tissue; they may also be something like a cyst or a benign lump, but this is why it is important to perform a monthly breast self-exam because you will get to know your breasts and know what lumps you have and what lumps may be new or growing. If there is a hard lump that is not moving or growing, a further checkup is required. A red or hot swollen breast is a common sign of infection such as mastitis from breastfeeding or from other skin changes such as eczema [24]. Further assessments are required if there is a color change due to the abnormal flow of lymphatic fluid [25]. A growing or enlarged vein on the breast can appear in response to circulatory issues, breastfeeding, or weight changes [16]. However, if a vein enlarges with other signs, such as redness or swelling, it may be a sign of breast cancer. In relation to sores on your skin, breast cancer can cause tissue changes from the inside to outside and can cause sores or wounds on the breast. You may also feel a lump with the sore if you have no known reason for a sore. This could be a sign of breast cancer, also known as Port Orange, when the skin of your breast appears dimpled, like how the peel of an orange looks; this can be a sign of breast cancer [26]. It is caused by breast swelling which makes the hair follicles appear dimpled on the skin. In relation to unexpected nipple discharge, nipple discharge is commonly related to developing breasts, infection, cysts, pregnancy, and breastfeeding, but if any of these common issues are not the cause, it can be a sign of breast cancer [16]. Itching of the breast can be a typical sign of skin changes, dryness, or even lotions or body washes that cause irritation, but if it does not go away with product changes or skin care, it may be a sign of breast cancer [27]. So, those are the main symptoms of breast cancer. Cancerous cells can develop at any age, irrespective of gender.

Screening techniques for breast cancer can help patients to detect cancer at the initial stage [28–30]. Screening can be performed in two ways: BSE and clinical breast examination (CBE) in which medical professionals examine the chest physically. In CBE, conventional breast screening methods such as mammography, ultrasound, positron emission tomography (PET), and computed tomography (CT) are employed. Mammography has a high sensitivity of 97% [31], a specificity of 64.5% [31], a positive predictive value of 89% [31], a negative predictive value of 90.9% [31], and diagnostic accuracy of 89.3% [31], but this technique has several limitations: ionizing radiation is used [32], the density of the breast affects the sensitivity and specificity [32], limited dynamic range, low-contrast and grainy image due to the sensitivity drops with tissue [31], etc. Ultrasound is not suitable for bony structure images even though this method has a sensitivity of 80% [33] and a specificity of 88.4% [33]. Ultrasound provides a low-resolution image, and an experienced operator

is required during the examination [32]. People also consider PET for screening of breast tumors. The sensitivity of PET is 68% for a tumor less than 2 cm [34] and 92% for a tumor of 2 cm to 5 cm [35], and the overall accuracy for detecting in situ carcinomas is low (sensitivity: 2–25%) [34]. PET has some disadvantages; viz., this method is an expensive method, uses ionized radiation [32], has limited resolution and slow imaging time [32], etc. CT scanning is an effective method for the early diagnosis of breast cancer due to its sensitivity (84.21%), specificity (99.3%), and accuracy (98.68%), but still, this method has limitations, i.e., successive CT scanning of radiosensitive organs like the breast can cause long-term effects [36–39]. When experts can observe initial symptoms associated with breast cancer in patients, they suggest applying an X-ray to the breast to collect an X-ray image for further observation and determination of breast cancer [30]. Sometimes, fluid-filled lymph is cancerous and is considered less dangerous than solid lymph. If there is a signature of fluid-filled or solid lymph, the breast is considered for an ultrasound checkup for further confirmation of breast cancer fill [33]. On the other hand, some specific cells from a particular regime associated with breast cancer can be removed by breast biopsy [36–39]. In that case, breast tissue is monitored under a microscope. Further investigation is performed to observe different growth factors associated with breast cancer, viz., estrogen receptor (ER) [40–42], progesterone receptor (PR) [41], human epidermal growth factor receptor (HER) 2 (HER2) [43], triple-negative breast cancer (TNBC) [44], etc. The test results associated with a biopsy can help doctors design a patient's treatment. However, all the methods mentioned above present several difficulties in generalized applications. This review article will focus on magnetic resonance imaging, microwave sensors [45], and different biosensors [46–48] in terms of mechanism, sensitivity, limit of detection, linear range, complexity, and availability for detecting early-stage breast cancer.

The rest of the paper is organized as follows: Section 2 presents different biomarkers associated with breast cancer. Section 3 highlights nuclear magnetic resonance imaging. Section 4 discusses microwave sensors for breast cancer. Section 5 explains biosensors for breast cancer. Section 6 represents challenges for existing techniques and prospects. Section 7 provides a brief conclusion.

2. Breast Cancer Biomarkers

Biomarkers are biological markers [49–51]. They are signals from the body that help doctors understand the impact of cancer cells on other cells. The production of biomarkers can be due to cancer itself or other cells in the body corresponding to the response of cancer cells. The information gained from biomarker testing allows doctors to create a personalized treatment plan based on the characteristics of a person's cancer [49–52]. Doctors investigate different biomarkers to evaluate cancer. Biomarkers may confirm the presence of cancer, diagnose cancer, identify the chance of a condition becoming more serious, be prognostic, or predict how a person might respond to a drug or treatment [49–51]. Biomarkers can help doctors to develop a personalized treatment plan. Biomarker testing allows the healthcare team to learn about a person's cancer. This can also be called molecular profiling, genomic testing, testing for mutations, etc. Doctors utilize biomarker test results to determine the traits of a cancer. Some biomarkers measure complex activity that occurs in the body related to the body's genes, proteins, and hormones [49-51]. Biomarkers can also measure genetic material from cancer cells in the bloodstream. Tests for these biomarkers are more specialized, and different cancer types are detected by other biomarkers [49–51]. For example, biomarkers for colorectal cancer are different from biomarkers for lung cancer. The most used breast tumor markers are ERs [52], PRs [52], HER2, TNBC [44], etc.

2.1. Estrogen Receptor

Estradiol circulates throughout the body in vascular circulation and freely diffuses through the cell's plasma membrane. Estrogen interacts with the ER in the plasma membrane, cytoplasm, and nucleus. Once estrogen binds to the receptor, it interacts with the immense heat shock protein HSP90 [53]. HSP90 dissociates, allowing the ER to undergo

a conformational change and dimerization, activating specific transcriptional sites on the receptor. Both activating function (AF) 1 and AF2 are activated in the ER. The dimerization ER then translocates into the nucleus and binds to the estrogen response element or ERE of DNA [53]. Co-activators and other transcription factors are recruited to the site, leading to this transcription of several estrogen-sensitive growth-promotion-associated genes via RNA polymerase. This causes further cancer cell division. Therapies that inhibit estrogen-estrogen receptor interactions can inhibit tumor growth [53].

2.2. Progesterone Receptor

PR is considered a hormone receptor and is governed by the ER. The interaction of PR and chromatin changes the binding position of chromatin and ER [54]. The expression of PR is dependent on estrogen. PR has been considered a prognostic biomarker associated with hormone-positive breast cancer. High expression of PR is related to tumors with a better baseline prognosis.

2.3. Human Epidermal Growth Factor Receptor 2

HER2 is a protein found on the surface of cells, regulating cell growth by transmitting signals from outside the cell to the nucleus inside the cell. Overexpression or an excess of HER2 receptors on a breast tumor cell or multiple copies of the HER2 gene in the cell's nucleus could lead to a particularly aggressive type of breast cancer [55]. This cancer became known as HER2-positive breast cancer. HER2 can impact a network of messages or signals inside the cell that helps tell the cell to grow, divide, or stay alive. This network is called the HER pathway [55]. Preclinical studies have shown that the HER pathway is overly active in HER-positive cancer. Too many signals running through the HER pathway can lead to uncontrolled cell growth and potentially contribute to cancer development. Receptors are a type of protein that exists on the surface of both standard and cancerous cells and are responsible for breaking signals from outside the compartment into the cell. Each receptor must dimerize with another HER receptor to send signals along the pathway [55]. Receptor pairing is thought to be critical in tumor growth and survival. However, any HER2 receptor can potentially contribute to human development and survival. The HER2 receptor may play the most significant role in cancer. HER2-containing receptor pairs lead to stronger signals through the HER pathway compared to receptor pairs that do not contain HER2. For example, the pairing of HER2 and HER3 is thought to trigger the most robust signaling of all receptor combinations [55]. HER2-positive breast cancer is often considered aggressive because it grows and spreads quickly and has a poor prognosis. HER2 overexpression affects about 25 percent of people with breast cancer.

2.4. Triple-Negative Breast Cancer (TNBC)

The lack of ERs, PRs, and HER2 expression is known as TNBC. TNBC is also associated with mutations in the BRCA1 gene, which is a gene linked to hereditary breast cancer. Many cells signaling pathways are dysregulated in TNBCs, such as the canonical WNT/betacatenin signaling pathway, which controls many cellular processes like multiplication, movement, and specialization. When interrupted, this process can lead to changes in how cells replicate, leading to tumor formation. In TNBCs, the WNT/beta-catenin pathway is over-activated. When WNT binds to frizzled receptors, another protein called AXIN becomes activated too. AXIN usually is part of the destruction complex, a collection of various proteins and kinases. When this pathway is inactive, the destruction complex prevents beta-catenin from accumulating in the cell by binding to it. However, without AXIN, the destruction complex cannot bind to beta-catenin. This allows beta-catenin to collect and eventually move into the nucleus, activating other co-activator molecules. These co-activators are responsible for triggering a group of oncogenes, genes involved in tumor formation. The proteins these oncogenes produce give rise to the proliferative, invasive, and metastatic abilities characteristic of TNBC cells. Like many diseases, the treatment of TNBC is tailored to the individual diagnosed. Standard treatment methods

include surgery, chemotherapy, radiation, and immunotherapy, depending on the stage of cancer development.

There are some nucleic acids such as circulating RNA (circRNA) [56], microRNA, long noncoding RNAs (lncRNAs), circulating tumor DNA (ctDNA), circulating cell-free DNA (ccfDNA), BRAC1, and BRAC2 which have been considered as cancer biomarkers [32]. Some proteins are also considered biomarkers, viz., CD24, CD44, MUC1, etc. Sometimes cell segregation can be established inside the tumor, and several cells can detach from the tumor. These detached cells can enter the blood system and are called circulation tumor cells (CTs) [52]. The level of CTs can be considered to determine the stage of metastatic cancer, and associated individual treatment can be provided to the patient. Some exosomes are regarded as a signature of cancer. However, ER, PR, HER2, and TNBC are commonly considered biomarkers for diagnosis and treatment plans.

3. Nuclear Magnetic Resonance Imaging for Breast Cancer Detection

Nuclear magnetic resonance imaging (NMRI) is one of the most modern approaches for observing the details of any human organ [57–59]. A breast NMRI is an advanced imaging examination used to help the radiologist or clinician screen for breast cancer in certain high-risk populations. It is also commonly used for further evaluation of a known or suspected breast cancer. NMRI is no different than the equipment frequently used to evaluate the brain, the spine, or the joints. The only difference is in the way it is performed. For breast NMRI, the patient lies face down on the NMRI table onto a padded breast coil. Following the intravenous injection of a liquid called gadolinium or contrast, a series of highly specialized images are then obtained. The appeal of breast NMRI is that it is the most sensitive examination. That means its ability to find small early cancers is highly efficient in women with a strong family history or other genetic risk factors. We can use breast NMRI to screen for cancer before any symptoms appear. MRI can also be used prior to treatment following a diagnosis of breast cancer to map out the area of disease more accurately for the surgeons. NMRI has shown a high sensitivity of up to 93% and a high specificity of up to 97.2% [60–62].

4. Microwave Sensors for Breast Cancer Detection

Several research groups worldwide have been studying the use of microwaves for medical applications, especially for breast cancer screening [63–69]. Microwaves are high-frequency, non-ionizing electromagnetic radiation used in daily devices like mobile phones [68,70]. Low-power microwave technology is currently easily accessible and low-cost and poses no health risk. Complex permittivity is one of the material properties that dictates response to electromagnetic waves [71–73]. It measures how molecules dynamically deform when subjected to alternating applied electromagnetic fields. Different tissues, forming the breast's internal structures, present complex permittivity [74].

If the contrast between the permittivity of various materials is high enough, this variation in permittivity causes the incident electromagnetic wave to reflect at the boundary. The dielectric constant of a material describes the dielectric characteristics of the material. The value of the dielectric constant associated with a specific material can represent the ability to absorb energy from an electric field. The smaller the dielectric constant value, the less energy is absorbed from an electric field.

Radar-based microwave imaging (RMI) systems provide the fastest imaging inversion times. To reconstruct the image, several antennas are distributed around the breast phantom (please see the left panel of Figure 1). A low-power microwave signal is transmitted from each antenna, and the system registers and processes all the picked-up reflections [68]. The next step is to acquire the electromagnetic signals reflected from the breast for all combinations of antennas [70]. The process is repeated for different antenna heights. The medical doctor can view the reconstructed image after the signal acquisition (see the right panel of Figure 1). The algorithm computes the image for the whole volume, which can then be sliced into cuts for precise examination. Several authors considered microwave

imaging for the detection of breast cancer. In [64], the authors used nine Vivaldi antennas around the breast phantom to observe the tumor. In that case, one antenna was considered an electromagnetic wave transmitter, the remaining antennas were deemed scattered electromagnetic wave receivers, and the same process was considered for other antennas. The Microwave Radar-based Imaging Toolbox (MERIT) has been used to reconstruct an image from backscattered signals [75]. The results from this simulation suggested that this antenna array can detect a spherical tumor with a 5 mm diameter.



Figure 1. Distribution of antennas around the breast phantom (**left** panel) and detection of the tumor by considering backscattered signals (**right** panel) [64].

The breast phantom is subjected to external electromagnetic radiation, and then some portion of the energy associated with electromagnetic radiation is absorbed by the phantom. The amount of energy absorbed by the phantom is known as the specific absorption rate (SAR). This energy is responsible for raising the temperature of different phantom parts according to their dielectric properties [67]. Usually, the effective treatment area in which the temperature rises 7 to 10 degrees is considered tumor necrosis. In [67], the authors applied the microwave imaging method by considering a different number of tapered slot antennas around the breast phantom and observed that (a) the SAR at the center of the breast phantom increased with the number of antennas around the breast phantom and (b) the temperature of the tumor's regime increased by 8 to 10 degrees after considering 12 antennas. Researchers incorporated artificial intelligence for signal processing to identify patterns to classify tissues and tumors [76–79]. So, this RMI is substantially suitable for detecting breast cancer at the initial stage. Table 1 represents a comparison of different antenna elements and the detected tumor radius.

Table 1. Comparison of different antenna elements for detecting a breast tumor.

Reference	Antenna Type	Number of Antennas around the Breast Phantom	Detected Tumor Radius
[64]	Vivaldi	9	2.5 mm
[80]	Quasi Log Periodic	16	5 mm
[81]	Monopole Microstrip	1	3 mm
[82]	Fractal	24	2 mm
[83]	Microstrip	1	10 mm

However, there are some complexities in generalized applications of RMI. First, the experimental breast phantom is static, while the real breast is dynamic. Second, designing an artificial breast phantom like the natural breast is challenging. There is some discrepancy in the dielectric properties of each layer of the artificial phantom compared to the real breast. Third, the time for the computation of relevant data is significantly longer than

that of other processes, and a powerful computer is required to complete the simulation. Recently, different magnetic nanoparticles have been considered along with the microwave imaging process to determine the status of a breast tumor.

Microwave-induced thermoacoustic imaging (MITI) is a modern imaging method in which electromagnetic microwave radiation is directly applied to the biological tissue to change the temperature of a specific portion of the tissue and for the generation of thermal waves by expanding and contracting a particular medium of tissue [84-86]. Finally, the specific tissue medium's expansion and contraction produces acoustic sound waves [87–89]. Several ultrasound transducers have been used to detect these acoustic sound waves and utilize them to construct an image of a specific tissue part. This method, which is a nonionizing and non-conducting method, is simple, safe, and suitable for application in medical science, viz., early detection and diagnosis of different cancers, brain imaging [86], fetus imaging during pregnancy, and cardiovascular imaging. In MITI, microwave pulses are applied to a particular part of the tissue, and the light-matter interaction produces different physical phenomena, viz., scattering, absorption, and reflection of the light [87]. But in this method, the absorption part is considered to observe the physical properties of the tissue [87–89]. So, when the microwave pulses are applied to the tissue, a local absorption of light is established, and the temperature of the associated regime increases, ultimately generating a sound wave [87]. Several ultrasonic sensors are used to detect the acoustic sound waves. This sound wave has different physical properties, viz., amplitude, frequency, and phase. The dielectric properties of the tissue with tumors and other abnormalities lead to a rigorous change in the amplitude, frequency, and phase of the sound wave [87]. So, the resulting images can provide detailed information about the tissue. This method has a unique high resolution, is easy to apply for differentiation of normal and abnormal tissue and is very useful for the early detection of different cancers [87–89].

In [84], the authors numerically studied the change in the physical properties of both tumors and other tissues in a breast phantom. Their simulation considered an operating frequency of 2.45 GHz and an isotropic electric field produced by changing the geometrical orientation of the waveguide and tank. For their simulation and associated results, they applied different conditions to run the simulation, such as changing the tumor's size, shape, and even orientation in the breast phantom and the microwave energy (ME) pulse width and power supply. The outcomes from their simulation are as follows: (a) the breast tumor experienced more ME absorption than another part of the phantom due to the higher dielectric properties of the tumor; (b) in comparison with breast fatty tissue, the tumor can absorb more ME and can increase in temperature, even showing large pressure gradient, which leads to generating an acoustic wave, which can also be observed corresponding to the tumor and associated regimes; (c) the location of tumor location in transitional tissue is more complex than the observation of tumor location in fatty tissue.

In [85], the authors considered the MITI system for detecting breast cancer in a threedimensional framework. In this study, authors applied ME to heat the breast phantom and simulated it with different conditions, viz., tumor size, shape, location, orientation, excitation power level, and power width. The authors found the following: (a) the ME absorption by the normal and abnormal tissues is not at the same level; (b) the condensed tumor tissues absorb more electromagnetic energy than the normal breast tissue; (c) the temperature of the tumor increases more in comparison to the other breast tissue; (d) the expansion rate and the pressure variation of the tumor are higher than those of the other tissues of the breast phantom; (e) the size of the tumor is associated with the amplitude of the acoustic waves, and a bigger tumor size produces a larger amplitude of the acoustic wave; (f) tumor shape can change the temperature of the tumor; (g) greater irradiation power can increase the temperature of both the tumor and other tissue; (h) the pulse width can increase the temperature as well as the magnitude of the acoustic wave; (i) finally, the simulation can detect a tumor with a radius 0.25 cm. The MITI technique is a powerful technique for the early detection of breast tumors.

5. Biosensors for Breast Cancer Detection

5.1. Surface Plasmon Resonance Imaging

The surface plasmon resonance (SPR) method has been used to study molecular interactions in real time [90–93]. The direction of light propagation is rigorously dependent on the medium, especially on the density of the medium in which light is propagated and the associated incident angle. When light propagates to the interface of two media, the light can undergo total internal reflection due to the specific incident angle and the density of both media [93]. For a plain polarized light undergoing total internal reflection, if a thin piece of metal is placed on the top, then some refracted light is absorbed. Due to this absorption, a dark zone appears in the refracted light. This absorption of light is carried out by the electrons present in the metal. Because of the absorption of energy, the electrons start oscillating. It is not a single electron, but a group of electrons collectively absorbing the energy of light [93]. These electrons undergoing oscillation because of energy absorption are called plasmons. The refractive index (RI) associated with the medium near the metal surface is responsible for changing the surface plasmon resonant angle, and a tiny change in RI can cause a significant change in the SPR angle [93]. SPR is widely used to study the interaction between two molecules, for example, the interaction between a ligand and its receptor. The interaction between a ligand and its receptor can be easily detected just by measuring the change in SPR angle. Biomolecules on a surface interaction between DNA and RNA can also be detected [93]. Figure 2 represents the experimental setup of SPR.



Figure 2. SPR experimental setup [94].

In [90], the authors developed an SPR immunosensor, in which gold nanoparticles were used to amplify the signal, to detect the ErbB2 biomarker associated with breast cancer, and they observed the following: (a) a limit of detection (LOD) of 180 pg mL $^{-1}$; (b) from breast cancer cell line graphs, it was shown that for different levels of ErbB2 (MDA-MB-436, SK-BR-3, MCF-7, etc.), the SK-BR3 cell line contains approximately 10 times more copies of the ErbB2 gene than both MDA-MB-436 and MCF-7 cell lines. In [91], the concentration of a carcinoembryonic antigen was determined by employing an SPR immunosensor, and the findings were as follows: (a) $3 \mu L$ of serum or plasma sample is required; (b) the LOD is 0.12 ng mL⁻¹, and the quantification limit is 0.40 ng mL⁻¹; (c) a linear response was observed from 0.40 ng mL⁻¹ to 20 ng mL⁻¹. This method is substantially suitable for the detection of cancer biomarkers. In [95], a photonic crystal fiber (PCF) biosensor based on SPR was developed, and the authors observed a high optical sensitivity of 6214.28 nm/RIU for the quasi-transverse magnetic mode and an optical sensitivity of 6000 nm/RIU for the quasi-transverse magnetic mode for breast cancer cells. In [96], the authors designed an SPR-based PCF biosensor to detect a cancer biomarker and found that (a) the refractive index (RI) varied from 1.392 to 1.401 for six different types of cancerous cells and (b) the maximum sensitivity was 7142.86 nm/RIU for MCF-7.

5.2. Colorimetric Biosensor

A colorimetric biosensor can detect the existence of different biomolecules or chemical compounds by using the concept of color change in a solution [97–99]. In a real experiment, a biosensor contains several sensing elements that can interact simultaneously with the specific analyte, viz., by breaking chemical bonds, generating new bonds, or any chemical reaction that ultimately leads to a color change in the solution or material, and these results are converted by a signal transducer into measurable data.

In [97], the authors consider HERCEPTIN as an antibody for detecting the HER2/neu receptor, one of the most remarkable biomarkers for breast cancer. For this purpose, the authors constructed a micro-calorimeter by considering micro-electromechanical technology and observed that (a) there was a peak output voltage due to the binding reaction of HERCEPTIN with HER2/neu receptor as well as the temperature difference; (b) the reaction was developed only in hot junctions; (c) the output voltage linearly (nearly) increased with HERCEPTIN quantities; (d) at the edge of the cell membrane, the intensity was dramatically increased, which means the antibody HERCEPTIN developed bonding near the cell membrane of the biomarker. In [98], the authors constructed a sandwich-type Bi2Se3-AuNPs calorimetric biosensor which has a high LOD of 10^{-18} M and a dynamic range of 10^{-18} M– 10^{-12} M. This calorimetric biosensor was developed under consideration of a three-step multiple signal amplification strategy for early detection of the BRCA1 mutation.

5.3. Surface-Enhanced Raman Spectroscopy

The Raman spectroscopy (RS) [99] technique has been employed to study the intrinsic properties of molecules and molecular mixtures in a liquid, slurry, paste, or solid phase. The vibrational transition of molecules has been considered to examine the structural properties of molecules by considering Raman spectroscopy, and a similar method has also been considered for infrared spectroscopy (IR) [100]. The basic difference between RS and IR is that RS is related to scattering while IR is related to absorption [101]. There are two types of scattering, Rayleigh, and Raman [101]. Rayleigh is elastic scattered energy, meaning the frequency of the scattered photon is the same as the excitation photon frequency. Rayleigh scattering is inadequate for finding detailed information about the chemical composition of different molecules. Ramen scattering, however, is inelastic photon scattering and is adequate for finding detailed information about the chemical composition of other molecules. RS requires two steps to occur for a molecule to Raman scatter. In the first step, the excitation-source photons excite the molecule into a virtual energy state. In the second step, the molecule relaxes by releasing scattered photons to a ground state. RS is rare, with only 1 in 10 million scattered photons being Raman scattered. Current technological advancements associated with solid-state lasers, gratings, and detectors make RS valuable for identifying and monitoring compounds in a reaction or crystallization. RS can be categorized in two ways. The first is Stokes RS, where scattered photons have higher energy than the incident exciting photons. The second is anti-Stokes RS, where scattered photons have lower energy than the incident exciting photons. Stokes RS generally has been considered for designing current Raman instruments. The interaction between the laser and the structure of the different chemical bonds among other atoms is responsible for changing the frequency of the incident photon. RS can address the specific frequency changes after the interaction of the incident laser with the chemical bond of the molecules and has been applied to recognize the bond structure of different molecules. The graph of Raman shift versus intensity associated with different chemical bonds among the molecules can express the deviation of the frequency of the scattered photon from the incident photon. The intensity associated with the Raman signal increases with the increase in the value of a chemical bond number. So, RS is a quantitative treatment for the chemical bond number. Additionally, when coupling a probe to a Raman system, measuring these bonds and their changes in real time is possible. This is valuable for understanding reaction characteristics such as kinetics initiation, endpoint, intermediates, crystal form, molecular backbone, and



mechanical information. The Raman scattering mechanism and the outcomes of experiment can be seen in Figure 3.

Figure 3. Detection of target microRNA-21 using SERS [102].

In [100], the surface-enhanced Raman scattering (SERS) technique was utilized to determine the existence of breast cancer cells/stem cells. For their experiment, the authors considered two material types, namely carbon and gold nanoparticles. The concentration of $100 \ \mu g/1 \times 10^4$ exhibited a 5.4 increment for breast cancer cells and a 4.8-fold increment for stem cells associated with breast cancer. In [103], the authors studied the heterogeneity of HER2 associated with breast cancer. In that case, they observed four different cell lines, viz., SKBR3, MCF-7, T47D, and MDA-MB-231, from high to low HER2 expression levels. In [103], the authors suggested a functionalized silver nanostructured plate with the SERS technique, and this platform could be used for cells with different HER2 expression levels and is mostly based on the identification of the most prominent Raman peak at 1002/cm.

5.4. Electrochemiluminescence Biosensor

Electrochemiluminescence (ECL) is a technique in which a specific chemical reaction is triggered by applying an electrical current resulting in light emission [104–106]. This process has been used in various fields, including analytical chemistry, clinical diagnostics, and biosensors. Biosensors are analytical devices that utilize biological molecules or living organisms to detect and measure different analytes [104–106]. These devices have numerous applications, including medical diagnosis, environmental monitoring, and food safety. The use of ECL in biosensors offers several advantages over other detection techniques. Firstly, ECL has high sensitivity (i.e., 0.1 fM to 5 Pm [107]) and can detect low concentrations of analytes with high accuracy. Secondly, ECL is a fast and reliable method providing real-time results with minimal sample preparation. Lastly, ECL has a wide dynamic range and can detect a broad range of analytes, making it an ideal detection method for many biosensor applications. One of the most common biosensor applications of ECL is in the detection of biomolecules such as proteins, DNA, and antibodies. In this method, the biosensor is designed to capture the biomolecule of interest using a specific recognition element such as an antibody [104].

The captured biomolecule is labeled with an ECL probe, emitting light when an electrical current is applied. The amount of light emitted is proportional to the concentration of the biomolecule, allowing for quantitative analysis. ECL biosensors have numerous applications in clinical diagnostics, including detecting infectious diseases, cancer biomarkers, and genetic disorders. These biosensors offer a compassionate and accurate detection method that can be performed rapidly with minimal sample preparation, making them ideal for point-of-care testing. In addition to clinical diagnostics, ECL biosensors are also used in environmental monitoring to detect contaminants such as heavy metals, pesticides, and toxic chemicals. These biosensors offer a reliable and cost-effective method for monitoring ecological pollutants, allowing for early detection and mitigation of potential hazards. In conclusion, ECL is a powerful technique that has revolutionized biosensor applications. ECL biosensors offer a sensitive and accurate detection method that can be performed rapidly with minimal sample preparation.

In [108], the authors designed an ECL biosensor by considering a silica-based mesoporous material for detecting SKBR-3 cells. The authors observed from the experimental setup that (a) the lower limit of quantitation was 20 cells/mL, (b) the linear dynamic range was 20 to 2000 cells/mL, (c) the experimental setup was able to determine the concentration of SKBR3 cells, and (d) MCF-7 and MDA-MB-231 cancer cell lines were determined. In [109], a sensing platform was designed by considering gold-nanoparticle-based carbon nitride, and this arrangement of a potential from -1.2 V to 0 V is associated with ECL in the presence of peroxydisulfate. The outcomes from this experimental setup are as follows: (a) the linear range was found to be from 10 fg mL⁻¹ to 100 ng mL⁻¹; (b) the LOD was 0.2 fg mL⁻¹; (c) the ECL emission experienced a peak at 550 nm; (d) a stable and amplified output response was observed. This ECL method is suitable for specifying cancer biomarkers.

5.5. Quartz Crystal Microbalance Biosensor

A crystal oscillator works on the principle of the inverse piezoelectric effect and is made up of piezoelectric material [110]. When an external voltage is applied with a specific frequency to a particular material, it produces mechanical deformation, and this material starts vibrating at the same applied frequency, which is defined as the inverse piezoelectric effect. Conversely, if we apply an external force to this piezoelectric material, it generates a voltage across the two terminals [110]. So, if we mechanically force them to vibrate at a specific frequency, they can develop an alternating current (AC) signal with the same frequency. This effect is known as the piezoelectric effect, and a material that shows this effect is known as a piezoelectric material. Different naturally occurring crystals like Rochelle salt, quartz, and tourmaline have these piezoelectric properties. Among these materials, Rochelle salt has the maximum piezoelectric activity, which means that for a given applied voltage, it generates the maximum vibration. But mechanically, this material is weak and can break very easily. Tourmaline has the most minor piezoelectric activity but is strong enough to resist breaking [110].

In [111], a CD44-biosensor was designed by following the quartz crystal microbalance (QCM) approach for breast cancer detection under consideration of different films. The experimental results are as follows: (a) the LOD for MDA-MB-231 (M231) cells was 300 mL⁻¹ while the LOD for MCF-7 cells was 1000 mL⁻¹; (b) the expression level of CD44 on M231 cells was double that on MCF-7 cells; (c) the M231 cells were less stiff than the MCF-7 cells; (d) the metastatic potential of cancer cells was rapidly detected using this method. In [112], authors applied QCM sensors to detect the biomarker CA15-3 associated with breast cancer and observed the following: (a) the one-dimensional ZnO nanostructure was appropriate for biosensor application; (b) the QCM sensor showed good stability with a response time of 10 s; (c) QCM had high sensitivity (25.34 ± 0.67 Hz/scale (1 U/mL)); (d) a linear (0.99) pattern associated with CA15.3 was observed for a range of 0.5 to 26 U/mL. In [113], the authors considered QCM for monitoring breast cancer marker CA15-3 and found that QCM sensors had high sensitivity (26. 303 ± 1.139 Hz/scale, 1 U/mL) and good linearity (0.960 ± 0.013) in the 0.5 U/mL and 100 U/mL concentration range of CA15-3.

5.6. Fluorescence Biosensor

If an external light is applied to fluorophores or fluorescent molecules, then the fluorophores or fluorescent molecules absorb this external light and are internally excited [114]. The excited fluorophores jump to higher unstable energy states [115]. Most of the time, the higher energy states of any system are inconsistent, and the fluorophores tend to return to the most stable lower energy states and become stable by releasing the excited energy. When the excited fluorophores tend to move the original lower stable energy states from the higher unstable conditions, the exciting energy is emitted by a fluorophore with a different color from the original absorbing light's color. The process of absorption, excitation, jumping to higher unstable states, and becoming stable after releasing and again jumping to the lower stable energy states is known as fluorescence. In fluorescence experiments, laser light is applied to fluorophores to excite them in a correct specific range [115]. After absorbing the energy from the laser light, these fluorophores jump to the higher energy states and again to the lower stable energy states. Finally, the detector detects the corresponding emitted light from the fluorophores [115]. In biological experiments, fluorescent dyes or antibodies have been used to detect the structure and processes of proteins, tissues, organisms, etc. The existence and number of fluorescent molecules can be determined using microscopy or flow cytometry techniques [115]. The fluorescence method is regularly considered due to its ultra-high sensitivity and process producing a quantitative result. DAPI, which has been used to detect DNA, and ethidium bromide, which has been used to intercalate DNA in agarose gels, are common fluorescent molecules. Green fluorescent protein can be found in jellyfish. A biological response has been converted to a measurable signal by a biosensor by combining enzymes or antibodies with a transducer. In that case, fluorescence molecules are considered as the transducer in fluorescence biosensors to produce a measurable signal.

In [116], the authors developed a complex 3D polymeric network, called a hydrogel, by combining carbon dots for early detection of cancer cells. For this purpose, the authors fabricated a nanocomposite hydrogel and functionalized it with an ssDNA probe to detect human breast cancer MCF-7 cell lines. In that case, they considered the DNA hydrogel bioassay strategy. They observed fluorescence spectra for three nanocomposite hydrogels (viz., GA-CDs-CH, NB-CDs-CH, and NB-CDs-CH) for a different concentration of microRNA-21. For a particular condition, the decreased fluorescence intensity, and the logarithm of microRNA-21 concentration provided a linear relationship at 0.1 to 125 fM for GA-CDs-CH and NB-CDs-CH hydrogel and 0.1 to 26.3 fM for B-CDs-CH hydrogel. These nanocomposite hydrogels are applicable for multicolor imaging of MCF-7 cancer cells. In [117], the authors employed the fluorescence emission method to detect biomarkers (microRNAs, miR-21, miR-195, and miR-155). They considered label- and enzyme-free methods and fabricated a fluorescence reporter based on DNA-templated copper nanoclusters to detect biomarkers. They found a linearity range of 500 nM to 3 μ M for a detection limit of 1.7pM. This fluorescent spectrum method is suitable for detecting microRNAs without any linkers and enzymes.

5.7. Flow Cytometry

In the flow cytometry method, the physical scattering process is considered to determine the size, shape, and properties of a single cell in a combination of different cells [118–120]. Initially, the sample is mixed with saline solution in the cytometer and allowed to pass through a narrowing channel which finally causes a single cell to live in a single droplet. When this droplet with a single cell passes through the laser pulse, the droplet/cell interacts with the laser pulse [118]. The point at which the cell interacts with the laser pulse is known as the "interrogation point". After the laser pulse interacts with the droplet/cell, it scatters in different directions. The portion of the laser that moves forward in focus after scattering is known as forward scatter (FSC) [118–120], and the portion of the laser that moves in the sideways direction after scattering is known as side scatter (SSC) [118]. Experimental setup of flow cytometry can be seen in Figure 4.

Forward scattering pulses provide information associated with the size of a cell, while sideways scattering pulses provide information associated with the shape and internal complexity of a cell [118]. The forward (sideways) scattering laser pulses are detected by detector 1 (detector 2) and converted into voltage pulses. These data are converted to a histogram plot according to the scattered light and cell number [118]. Researchers investigate both forward and sideways scattering data together and determine a cell's size, shape, and complexity [118–120].



Figure 4. Experimental setup of flow cytometry [121].

In [122], the authors applied the flow cytometric method to determine the number of cancer cells in peripheral blood. They studied the reactivity of several reagents, viz., anti-CK20, anti-CK7, anti-pan-CK, anti-CK8, anti-CK8/CK18, and anti-CK18, and found that this sensing system could detect a single cancer cell from 10⁶ to 10⁷ nucleated blood cells. In [123], authors saw different types of breast cancer markers, viz., ER, PR, HER-2/neu, (EGFR), and E-cadherin, by applying the flow cytometry method and reported the following: (a) a rigorous positive pattern had been observed for E-cadherin with an 85–92% marker expression; (b) in comparison with E-cadherin, a weakly positive expression pattern had been demonstrated by ER, PR, and HER-2/neu; (c) there was an inverse relation between HER-2/neu expression levels with MCF7 cell lines. So, flow cytometry is a powerful tool for detecting breast cancer in the primary stage.

5.8. Microfluidic Chip

Fluid can flow in two different ways, viz., laminar flow and turbulent flow. In laminar flow, the streams are parallel [124]. So, there is no mixing between the streams. Then there is turbulent flow, where various inertial forces cause the fluids to move randomly. Turbulent flows tend to affect reading accuracy [124]. In the micrometer regime, the liquid flow is to be considered as laminar flow. So, a complex pattern is produced by the joining points of two fluid streams in a microchannel, and diffusion is required to explain this.

On the other hand, the scaling law is a comprehensive way of describing the variation in the physical quantity as a function of the size of the system [124]. Microfluidics is a study associated with submicron fluidic structures, and this knowledge is applied to developing lab-on-chip or microfluidic chip technology [124], which combines different operations. All kinds of operations are completed in a micron-sized chip. Microfluidic chips use semiconductor-like micro-electromechanical processing technology to build a microfluidic system on the chip [124]. Different interconnected paths and liquid phase chambers exist in the chip, and the analysis process associated with the experiment is printed over the chip [124]. Biological samples with their solutions have been loaded, and different pumps and forms of electro-osmotic flow have been used to transmit the flow of the buffer in the chip to form a micro flow path. So, the flow experiences several successive reactions over the traveling trajectory. For accurate and precise sample analysis, different detection systems, along with analysis methods, have been employed in microfluidic chips [124]. A complete microfluidic system includes many components, mainly micr-pumps, microchips, microvalves, and detection units. A micropump is the power source for microfluidic transportation [124]. Microchips can also be classified according to different dimensions and according to the material of the chip. They can be divided into silicon, glass, organic polymer materials, and paper materials [124]. Based on the types of channel arrays, they are single-channel and multi-channel chips. According to the network shape, they can be divided into linear, spiral, curved snake, polygon, folded, etc.

In microfluidic technology, a small chip can integrate several steps associated with miniaturing and automation. There is high throughput since microfluidics can be designed as multiple channels, and multiple items can be tested in parallel on the same sample as needed. The detection time is significantly shortened, the efficiency is improved, and high-throughput detection is realized. The reaction unit cavity associated with the microfluidic chip is tiny. The reagent usage is far lower than that of conventional methods, and thus, the consumption of reagents is greatly reduced. The demand for sample sizes is small. Since the task is completed on a small chip, the sample size required to be tested is very small, so microliters or even nanoscale amounts could be enough [124].

Biological analysis based on microfluidics includes microfluidic-based analysis in cell mode and microfluidic-based exosome separation analysis [125]. Clinical diagnostic research based on microfluidics includes pathogen detection and analysis based on microfluidics. Cancer detection and analysis can be based on microfluidics. Food safety analysis, including detecting pesticide residuals, pathogenic bacteria, heavy metals, and food additives, has been based on microfluidics [126]. Environmental monitoring and analysis have been based on microfluidics detection methods including electrochemical, optical, and MS methods. Creative Biolabs is a leading customer service provider in cancer next-generation sequencing with a professional platform and rich experience. Integrated microfluidic devices could differentiate breast cancer patients from healthy controls with high sensitivity (90%) and specificity (>95%), making this strategy applicable to the analysis of clinical specimens [127].

In [128], the authors considered a glass substrate and polydimethylsiloxane for constructing a microfluidic device applicable to quantifying the circulating exosomes. They collected and measured the circulating EpCAM-positive exosomes from six breast cancer patients and three healthy people and observed that the level of EpCAM-positive exosomes significantly increased in breast cancer patients compared to healthy people. Personalized breast cancer therapy is associated with molecular classification by HER2. The authors considered nineteen breast cancer patients with HER2-positive exosomes. They utilized a microfluidic device to discover the secret HER2-positive exosomes in breast cancer cells. They observed that the level of HER2-positive exosomes in SK-BR-3 CM is significantly higher than that in both MCF7 and M231 CM. In [129], the author designed a detection platform for miRNA biomarkers by considering a microfluidic chip and observed that the detection limit was 1 pM and the time was 30 min. This microfluid chip approach is suitable for detecting breast cancer biomarkers and quantifying the amount of existence.

5.9. Molecularly Imprinted Polymer

A molecularly imprinted polymer (MIP) is a polymer that has been designed by applying the molecular imprinting technique, and MPI can generate cavities in the matrix of the polymer to attract a particular template molecule [130]. Manufacturing a nano-MIP involves controlled polymerization around a targeted choice, immobilized on solid-face support [131]. This is then packed as a column. The nano-MIP is made by first exposing the target to monomers that find their optimal binding positions on the target through self-assembly. Different solvents and cross-linking agents have been used to complete this step. In the pre-polymerization step, the distribution of the functional monomers around the template is precisely organized by the chemical properties and shape (see Figure 5).



Figure 5. Different steps of molecularly imprinted polymer preparation [132].

The monomers are fused together through polymerization by exposing the column to UV light, activating a catalyst that generates free radicals. The free radicals collide with the carbon-to-carbon double bonds in the monomers, initiating a polymerization chain reaction. Once sufficient polymerization has occurred to form the desired level of polymer formation around an individual template molecule, we stop polymerization. Stopping polymerization is simple. The UV lamp is turned off, and the living radical becomes an inhibitor, mopping up the free radicals. Each discrete single polymer particle or nano-MIP has one binding site. We then wash away the unreacted monomers and lower-affinity polymers. By controlling the stringency of the wash process, we regulate the affinity of the nano-MIPs, which are subsequently harvested. We can further modify the surface of the nano-MIPs retained on the column by attaching other compounds of interest, such as biotin or peg. This allows us to create the best possible MIPs for application in a new world of diagnostic solutions. In [133], the authors observed that the sensitivity of the ammonia sensor developed using the MIP-doped boric acid method was 0.243 Hz/ppm, which is three times greater than that of a sensor developed without the MIP method (i.e., 0.079 Hz/ppm).

In [134], carbohydrate antigen 15-3 (CA 15-3) was detected by employing the MIP strategy, and the following observations were made: (a) the redox peak current decreased with an increase in the value of the CA 15-3 concentration; (b) LOD was observed 0.10 U mL⁻¹; (c) the MIP biosensor detected the CA 15-3 associated with breast cancer below the clinically recognized value. In [131], the author considered the MIP method to observe the susceptible BRCA-1 gene associated with breast cancer and reported that the suggested biosensor maintained a wide linear range from 10 fM to 100 nM and had an LOD of 2.53 fM.

5.10. Field-Effect Biosensor

Semiconductor-based biosensors (SBBs) can be directly applied to extract information associated with different cancers from biological experiments [135,136]. These SBBs have been designed to observe a sample's electrical property variation. The results that are found from these experiments are used to determine the status of the cancer and diagnose the condition. A field-effect transistor (FET) is a semiconductor device with three terminals: source, gate, and drain. The FET controls the conductivity between the drain and source by considering the electric field or the application of a voltage to the gate [135]. A field-effect biosensor (FEB) is a device manufactured by following the principle of the FET that can detect the existence of biological molecules, such as proteins or DNA, in a sample [137–139]. Currently, several FEBs are available, viz., ion-sensitive field-effect transistors, metal-oxide-semiconductor field-effect transistors, organic field-effect transistors, and nanowire field-effect transistors [140]. Currently, ultra-sensitive nanowires can be utilized for developing lab-on-a-chip devices. The epitaxial method has been employed by considering nanowires for designing three-five SBBs. In that case, a metal droplet is deposited over the surface of the semiconductor, and nanowires vertically grow from the metal droplets [140–142]. The nanowire behaves like a junction, and due to the ultrasensitive properties of nanowires, it can easily capture and transmit the optical and electrical signals to the next portion of the device. Nanowires are so thin that the free charges exist at the nanowires' surface [140-142]. So, when target molecules contact the nanowires, the surface free charges from the nanowires bond to the target molecules, and a change in the nanowires' surface charge density is observed. So, the interaction between the nanowires and target molecules can change the surface charge density of the nanowires, and this change in the charge density can indicate the existence of different biomarkers [140–142]. So, the idea of FET is an ultra-sensitive approach to recognizing the existence of specific biomarkers, and a carbon nanotube (CNT) FET device with high sensitivity down to the fM range can be regenerated in an acidic environment, enabling the reusability of the sensor device [140-142].

In [135], the author considered a DNA-functionalized CNT-based FET biosensor in which gold nanoparticles are considered linkers to anchor probe molecules to detect breast cancer exosomal miRNA. Initially, exosomes are collected from the breast, and specific

miR21 is extracted from the exosomes. Hybridizing the target miRNA and immobilized DNA probe establishes a current flow. So, before and after the hybridization, the variation in the current flow can predict the existence of exosomal miRNA21. The CNT-based FET biosensor can directly convert the hybridization of biomolecules with a sensing interface into an electrical signal.

In [136], a single-nanotube FET array was considered to detect a change in the electrical signals due to the development of a bond between the receptor proteins and their corresponding antibodies. In that antibody device experiment, the authors considered two antibodies (IGF1R and Her2) at different levels. They established an interaction with intact MCF7 and BT474 collected from human breast cancer cells, and a 60% conductivity decrease was observed when BT474 or MCF7 breast cancer cells interacted with IGF1R and Her2 antibodies. The antigen-antibody interaction changed the system's free energy, increased the band gap, and decreased the system's conductivity. A single cell with a diameter 1000 times smaller than a cancer cell produced a bond with a single-nanotube FET, changing the conductivity.

5.11. Electrochemical Impedance Spectroscopy

Electrochemical impedance spectroscopy (EIS) applies an external sinusoidal potential/current to an electrochemical system and measures the corresponding outcome sinusoidal potential/current [143–145]. The applied potential has several features, viz., amplitude, angular frequency, and time dependency, and the corresponding output current has similar characteristics. However, the output current experiences a phase shift from the input external potential [144]. So, to reiterate, the input and output signals should have the same angular frequency, but a phase shift or a phase angle may offset the output signal. A complete EIS experiment applies a sinusoidal potential with different frequencies instead of a single frequency, and corresponding output currents are measured. So, a frequency spectrum is observed instead of a single frequency. Fourier transform has been utilized to convert all input and output signals from the time domain to the frequency domain. The experimental setup is a frequency-dependent system, so the impedance is calculated by calculating the ratio of the frequency-dependent potential over the frequency-dependent current [145]. The phase angle and impedance are a function of frequency, and the associated plot is known as a body plot. Sometimes, a Nyquist plot is used to show the real and imaginary impedance. The time scale is a critical parameter in electrochemical processes [145], and in most cases, the time scale associated with different mechanisms cannot be recognized globally but can be recognized locally. For example, double-layer charging can occur in a few microseconds while diffusion can happen in a time scale of hundreds of milliseconds [144]. These two mechanisms have two distinct time scales for completion, but in electrochemistry, these two-time scales are not distinguishable.

For example, in a DC voltammetry experiment, a step potential or a linear sweep is applied, and the corresponding measured current is the combination of different electrochemical processes occurring within a specific time. In that case, it is not easy to differentiate the current related to the double-layer charging or diffusion. So, DC voltammetry cannot provide a detailed picture of different mechanisms and their associated development time. On the other hand, EIS, in which oscillating perturbations are applied and corresponding currents are measured, can provide detailed pictures of individual internal mechanisms of the electrochemical reaction, and this EIS method is able to discern different electrochemical phenomena because they occur at different frequencies. EIS is widely used for direct monitoring of antibody-antigen connected reactions, DNA hybridization, and enzyme reactions, and the sensitivity of EIS can reach up to μ g/mL to pg/mL [146,147].

In [143], the authors employed the impedance spectroscopy method and differentiated early-stage, invasive, and metastasized human breast cancer lines MCF-7, MDA-MB-231, and MDA-MB-435, respectively, from the standard breast tissue cell line MCF-10A. They measured the impedance for both normal breast cells and three other cancerous cell lines over a frequency range of 100 Hz to 3.0 MHz and observed the following: (a) the magnitude

of the impedance decreased with frequency; (b) the phase of the impedance initially decreased and further increased with frequency; (c) there was a significant difference between the standard cell line (for both magnitude and phase of the impedance) and the other three cancerous cell lines; (d) the magnitude and phase of the impedance decreased (increased) following the order MCF-10A, MCF-7, MDA-MB-231, and MDA-MB-435. In [148], authors identified the spectroscopic patterns for three different pathologic stages, viz., the early stage (MCF-7), invasive phase (MDA-MB-231), and metastasis (SK-BR-3), by employing bioimpedance spectroscopy under the consideration of magnetic nanoparticles. In the alpha dispersion bandwidth, the magnitude of electrical impedance spectra maintains a substantial difference for different cancer cell lines. So, impedance spectroscopy is suitable for identifying low concentrations of cancer cells.

HER2 is considered one of the most important biomarkers for identifying the proliferation of breast cancer. In [149], authors designed an electrochemical immunosensor for analyzing the HER2 extracellular domain (ECD) in human serum, and a screen-printed carbon electrode was considered as a transducer surface in the presence of gold nanoparticles. The linear sweep voltammetry (LSV) method was applied to analyze the enzymatic reaction and the antibody-antigen interaction. They observe the following: (a) the quantification and the patient's response to the therapy of the biomarker were determined for a wide range of concentrations; (b) the dilute sample with buffer was sufficient for a high HER2 ECD concentration; (c) the LOD was determined at 4.4 ng/mL; (d) the cut-off value was 15 ng/mL, so the LOD, as well as the diagnosis of HER2, was easily determined for the patients.

5.12. Light-Based Breast Cancer Techniques

Light can interact with the matter of different refractive indices, and the reflection, absorption, polarization, and scattering of the light can be monitored to explore the characteristics of the tissue, cells, or biological materials [150]. We have already discussed different optical biosensors, viz., SPR, SERS, fluorescence, and flow cytometry, in which physical properties of light have been considered to detect breast cancer. Several authors considered the frequency-domain photon migration (FDPM) [151–153] method for detecting breast cancer. FDPM performs content-based analysis by deriving scattering and absorption coefficients from amplitude and phase differences in backscattered, broadband-modulated (10 MHz to 1 GHz) near-infrared light [154]. Terahertz (THz) imaging is another platform for detecting breast cancer at early stages. A THz wave (0.1–10 THz) has high sensitivity to water molecules [155], is non-ionizing [156], produces low power levels [156], and has good penetration capabilities and resolution. In [155], the author demonstrated THz reflection imaging and distinguished 10 µm heterogeneous tissue sections from cancer and noncancer breast tissue. In [156], TPS Spectra 3000 pulsed THz imaging and spectroscopy were considered, and it was found that tumor detection is accurate to depths of over 1 mm.

5.13. Analytical Techniques Based on Nanomaterials for the Detection of Breast Cancer Biomarkers

There are different nanomaterials like carbon nanotubes, graphene oxide, quantum dots, multifarious composites, and noble metal nanomaterials (Au NPs, Ag NPs) which have been considered as a part of different biosensors to improve the detection performance of biosensors [157]. Nanomaterials are hundreds of times smaller than human cells. So, the size and the quantum tunneling effect of the nanomaterials significantly improve the performance of biosensors. In [158], the authors constructed a CdS/ZnO nanorod array-based photoelectrochemical sensor to detect MCF-7 and determined the concentration from 50 to 1.0×10^6 cells/mL, and the LOD was found to be 10 cells/mL. In [159], the authors detected the breast cancer biomarker microRNA-21 (miR-21) by applying an ECL technique with a gold-silver nanocluster supramolecular network (AuAg NCs) and reported that the linear range was 10 aM–10 pM, and the LOD was 4.5 aM. Nanomaterials are also considered for SPR [160,161], SERS [162], and colorimetry [163–165], etc.

6. Challenges for Existing Techniques and Prospects

NMRI pictures can express a minimal change inside a medium. Sometimes doctors suggest the consideration of the breast under an MRI experimental setup and observe a small change associated with breast cancer to determine cancerous cells further. NMRI can successfully select a shift inside the breast even though the mammogram cannot do that [57–59]. NMRI is suitable for early breast cancer detection, but the availability of NMRI is challenging in different parts of world. Sometimes, patients feel uncomfortable with NMRI due to its relation to nuclear technology. Most of these methods are cost-effective, time-consuming, and unavailable worldwide. Again, this method is hard to manage for early detection of breast cancer or regular checkups for cancerous cells.

Microwave imaging is considered an effective method for early breast cancer detection. However, these methods have some limitations. The dielectric properties of the breast phantom are not precisely measured. The contrast between the tumor and other tissues' dielectric properties is minimal for detecting the existence of a tumor inside the breast specifically. Antennas are the most critical components in microwave imaging systems. The mutual coupling between these antennas also increases with the increasing antenna numbers of a microwave imaging system [45]. This mutual interaction of the antenna's radiation further leads to a low-resolution tumor position and size image. On the other hand, a lot of data are associated with different functions and orientations of the antennas. So, it is essential to utilize a specific method to determine the physical properties of individual data related to simulation. So, in that case, we again face a significant challenge in encapsulating extensive data and defining a specific binary decision corresponding to this large dataset. Several researchers applied the concept of artificial intelligence or neural networks to certify individual datasets and predict the future of breast cancer [166–174]. So, artificial intelligence is one of the most promising tools for generalizing the problem for individuals' datasets and determining the corresponding possibilities.

In MITI, the scattering between the incident electromagnetic wave and living tissue reduces the intensity of the incident light, and the absorption of electromagnetic waves by the living tissue sometimes causes deformation of the cells. On the other hand, the temperature variation and the sound wave associated with electromagnetic phenomena ultimately provide an indication of a tumor in a regime instead of a specific point.

SPR has many advantages; viz., it is a level-free method, and potential leveling artifacts are not required. It is a direct method because it measures the binding of the actual analyte. This method can provide information about the evolution of the experiment in real time. However, SPR has some disadvantages. SPR is designed for mass change, decreasing with the distance from the surface. The LOD is around 200 nanometers, and this method is also dependent on the material, especially gold and silver. Highly viscous materials cannot be considered for an experiment, and the sample must be homogeneous to be used in the investigation. Sample preparation and probe attachment to the metal surface are challenging.

A colorimetric biosensor is a fast, inexpensive, and portable detection device that can be used for the early detection of cancer biomarkers. However, this process has trouble determining the individual components of a mixture. Chemometric methods have been considered for encapsulating large datasets of red, green, and blue values [175]. Moreover, challenges are also associated with the reproducibility of imaging and printing.

SERS can be applied to find the information of a submicron system. SERS can determine the bonding information associated with different phases of materials. Generally, this experiment can be performed in ambient conditions and can study the dynamic process. Transparent or semi-transparent materials can be considered for SERS treatment within the visible spectrum for detailed bonding information. There is freedom in the experimental setup to visualize a wide range of in situ capabilities, viz., temperature, environmental control, strain, chemical monitoring, etc. However, SERS is not free from the disadvantages. A laser is applied to the sample in SERS treatment, and the exciting laser selection is essential for extracting the bonding information because the laser is also sometimes responsible for

19 of 30

degrading the material. RS is also a low-probability light interaction. Our ability to detect light has improved, and the limited detection of most materials is around 1%.

ECL is a non-radioactive, flexible, and fast process for detection. However, the complexities of applying this method are related to maintaining a favorable temperature and stable electrodes and the cross-interaction between the co-reactant and the luminophore. A quencher can be introduced further to amplify the intensity of ECL [176]. An ideal luminophore, which has a lower energy band gap, which means little energy is required to make an electron transition from the valance band to the conduction band, can be considered for ECL to improve the performance of ECL [176]. Modern quantum materials are used to design these highly efficient luminophores.

There are several advantages to considering a QCM biosensor. A QCM biosensor has stable, high-sensitivity, and level-free detection capability [177]. So, the possible interference associated with the actual binding process can be ignored in this approach [177]. However, a QCM biosensor has some limitations. First, if there is non-uniform mass distribution on the quartz crystal, this leads to a frequency discrepancy. Second, the solution leak enhances the capacitive background current. Third, a minimum time, the upper limit on the scan rate, is required to locate the resonant frequency. Fourth, the QCM frequency is linearly dependent on the temperature [178]. The temperature variation can change the experimental setup. Fifth, there is noise due to the vibrating nature of the crystal. The electrochemical signal must be separated from the exciting signal.

In most fluorescence biosensors, the fluorescence intensity is considered for detecting impurities. The instrument's condition and environmental factors can easily perturb the outcome fluorescence profile. In that case, the self-calibration of the two fluorescent emission bands has been addressed for ratiometric sensors to minimize the external effects on the fluorescence intensity profile [179,180]. In fluorescent resonance energy transfer (FRET) based ratiometric sensors, the ratio of two fluorescent emission bands is not affected by the concentration and fluctuation of the source [181]. Conjugated polymers are sometimes considered for delocalizing electrons and light harvesting, ultimately amplifying the fluorescence intensity output profile [181].

Flow cytometry is an optical setup to observe individual cell properties to differentiate abnormal cells from normal ones. In that case, the wavelength of the incident laser light is comparable to the size of the cell. Otherwise, the light-matter interaction will not provide detailed information about a single cell. On the other hand, this method can generate colossal data by considering a single cell, and this massive information can be categorized by employing the knowledge of machine learning [166–171] to determine the binary outcomes and corresponding possibilities.

Microfluidic biosensors are more compatible than non-microfluidic biosensors and can produce both quantitative and qualitative results within 3 h. The LOD for microfluidic biosensors is 3×10^3 CFU/mL, while the LOD for non-microfluidic biosensors is 3×10^4 CFU/mL. However, microfluidic biosensors have a smaller dimension and high surface-area-to-volume ratios, which ultimately leads to a lower flow rate and a high-pressure difference over a small distance.

The Debye shielding effect, which can be observed in any FET-based biosensor, is considered to reduce the LOD. The electrostatic field of the surface charges associated with biomolecules will vanish after the Debye screening length [182]. The number of accounts near the surface of the biomolecule and the same number of opposite directions also exist around the biomolecules. These opposite charges can materialize the electric field due to surface charges. In a buffer solution, the electrostatic field exponentially decreases and becomes zero after a particular distance, and this critical distance is known as Debye screening length. In FEB, the LOD is rigorously dependent on the Debye length limitation [182]. The sensitivity of an FET-based biosensor is significantly reliant on the strength of the ions.

Biosensors face several challenges in determining the status of a biomarker in a sample. For example, a lower concentration of biomarkers is one of the most critical factors. In that case, a lower biomarker concentration produces weak electrochemical interactions and associated weak signals, which most biosensors have difficulty detecting. In that case, several secondary antibodies or antibody-like substances are to be considered for amplifying the electrochemical signal [52]. On the other hand, the interference between different meanings generates different signals/noises. To resolve these complexities, several ideas have been considered in sensing techniques. Examples include the following: (a) quantum nanomaterials have been considered as an amplifier of the weak primary signals; (b) secondary antibodies can be introduced to amplify the raw signals; (c) secondary/indirect and larger indirect signals can be detected, and these secondary/indirect signals can be decomposed to observe the properties of primary/direct signals [52]. The sensitivity of a biosensor could be improved under the consideration of detection strategies for secondary/indirect signals. Background interference can be minimized if the sample undergoes a pretreatment. For

MIP, it can be considered that biomolecules have difficulty coping with harsh environments. Finally, different biomarkers can simultaneously exist in a single sample, but most biosensors can detect one at any time. Table 2 represents the linear range and the limit of detection of different biosensors for different biomarkers.

Type of Biosensor	Target	Linear Range	Limit of Detection	References
SPR	HER2	10^{-12} - 10^{-6} g/mL	9.3×10^{-9} /mL	[183]
	MCF-7	10^4 – 10^6 cells/mL	500 cells/mL	[184]
Calorimetric	BRCA1	1 fM-100 pM	0.34 fM	[99]
	BRCA1	$10^{-12} \text{ M} - 10^{-18} \text{ M}$	$10^{-18} { m M}$	[98]
SERS	CA153	-	0.0001 U/mL	[185]
ECL	HER2	0–900 pg/mL	20.4 pg/mL	[186]
	MCF-7	10–100 cells	10 cells	[187]
	CA19-9	0.0001–10 U/mL	0.000046 U/mL	[188]
	miRNA-21	100 aM–100 pM	19.05 aM	[189]
	CA19-9	0.0005–150 U/mL	0.0002 U/mL	[190]
	MUC1	1 fg/mL–1 ng/mL	0.62 fg/mL	[191]
	HER2	1 fM-1.0 nM	1 fM	[192]
QCM	MCF-7	10^2 – 10^5 cells/mL	430 cells/mL	[193]
	MCF-7	10^2 – 10^7 cells/mL	32 cells/mL	[194]
	MDA MB 231	-	12 cells/mL	[195]
Fluorescence	CA125	10^{-2} –127 U/mL	10^{-4} U/mL	[196]
	CYP1A1	-	1 pg/mL	[197]
	miRNA-21	1 pM–10 nM	55 fM	[198]
	miRNA-21	-	18.7 pM	[199]
Microfluidic	HER2	1 fM–100 nM	1 pM	[200]
	CA15-3	10–1000 μU/mL	6.0 μÛ/mL	[201]
	CEA	-	0.2 ng/mL	[202]
	Glypican-1	-	$10 \operatorname{exosome}/\mu L$	[203]
	PTK7	-	0.4 nM	[204]
MIP	CA 15-3	0.1–100 U/mL	0.10 U/mL	[134]
	HER2	1–200 ng/mL	0.43 ng/mL	[205]
	CA 15-3	5–50 U/mL	1.5 U/mL	[206]
Field effect	miRNA-155	0.1 fM-10 nM	0.03 fM	[207]
	HER2	100 pM–1.0 μM	1.0 pM	[208]
	EGFR	10 fM–10 nM	10 fM	[209]
	CA125	10^{-9} – 1.0 U/mL	$5 imes 10^{-10}~{ m U/mL}$	[210]
EIS	BRCA1	50 fM-1.0 nM	1.72 fM	[211]
	HER2	1 pg/mL–100 ng/mL	172 pg/mL	[212]
	miRNA-155	10 aM–1.0 nM	5.7 aM	[213]
	miRNA-21	10 fM–100 pM	4.3 fM	[214]
	EGFR	1 pg/mL–1 μg/mL	0.88 pg/mL	[215]
	HER2	0–4.0 μg/L	6.0 μg/L	[216]

Table 2. A summary of linear range and limit of detection of different biosensors.

Determining different biomarkers in a sample at a specific time is to be considered a primary challenge. In that case, several biosensors can be considered for detecting different biomarkers in the same sample at a specific time. This challenge can be resolved if several techniques are considered to design a complex biosensor that has high sensitivity, specificity, and a simple operation procedure and can simultaneously detect multiple targets. This complex biosensor can be designed under two considerations: (a) the strategy, by which biomarkers are to be detected, should include improving the sensitivity and enhancing the specificity, and (b) the detection equipment should be easy to operate and able to complete multi-target detection simultaneously. For developing a complex biosensor, the mechanism of electrochemical and optical biosensors can be combined with the mechanism of microfluidic chips to achieve better performance. Most of the microfluidic-chip-based biosensors are innovative in devices, while electrochemical and optical biosensors are more innovative in terms of detection strategies and materials.

7. Conclusions

Breast cancer is a global problem. Detection of breast cancer at an early stage is significantly important for designing individual treatment plans. Biomarkers that are the primary signatures of breast cancer and a detailed discussion about the biomarkers associated with breast cancer have been highlighted. There are two ways, i.e., BSE and CBE, by which breast cancer screening can be performed. This review article reported various approaches for BSE as well as different techniques (viz., mammography, ultrasound, PET, CT, NMRI, RMI, MITI, SPR, colorimetric, SERS, ECL, QCM, fluorescence, flow cytometry, microfluidic chip, MIP, field-effect, EIS, light-based, and nanomaterial-based analytic methods) associated with CBS for the early detection of biomarkers. Several parameters, i.e., sensitivity, specificity, accuracy, LOD, linear range, tumor radius, and temperature variation, have been considered for comparing different techniques. This review also addressed the existing challenges for detecting biomarkers, such as sensitivity, specificity, weak signal, background interference, and detecting different biomarkers at a specific time, and provided different solutions to resolve these challenges.

Author Contributions: Conceptualization, L.W. and N.A.C.; formal analysis, L.W., N.A.C., L.G. and M.K.; writing—original draft preparation, N.A.C.; writing—review and editing, L.W., N.A.C., L.G. and M.K.; project administration, L.W., L.G. and M.K. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: All data are presented inside the manuscript.

Acknowledgments: The authors are grateful to anonymous reviewers for their constructive suggestions which have significantly improved the quality of our manuscript.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Siegel, R.L.; Miller, K.D.; Jemal, A. Cancer Statistics, 2017. CA Cancer J. Clin. 2017, 67, 7–30. [CrossRef]
- Bray, F.; Ferlay, J.; Soerjomataram, I.; Siegel, R.L.; Torre, L.A.; Jemal, A. Global Cancer Statistics 2018: GLOBOCAN Estimates of Incidence and Mortality Worldwide for 36 Cancers in 185 Countries. CA Cancer J. Clin. 2018, 68, 394–424. [CrossRef] [PubMed]
- 3. Siegel, R.L.; Miller, K.D.; Jemal, A. Cancer Statistics, 2019. CA Cancer J. Clin. 2019, 69, 7–34. [CrossRef] [PubMed]
- 4. Rebner, M.; Raju, U. Noninvasive Breast Cancer. *Radiology* **1994**, *190*, 623–631. [CrossRef] [PubMed]
- Schnitt, S.J. Classification and Prognosis of Invasive Breast Cancer: From Morphology to Molecular Taxonomy. *Mod. Pathol.* 2010, 23, S60–S64. [CrossRef]
- Giordano, S.H.; Buzdar, A.U.; Smith, T.L.; Kau, S.-W.; Yang, Y.; Hortobagyi, G.N. Is Breast Cancer Survival Improving? Trends in Survival for Patients with Recurrent Breast Cancer Diagnosed from 1974 through 2000. *Cancer* 2004, 100, 44–52. [CrossRef]

- Loeser, A. The Insider's Guide to Metastatic Breast Cancer: A Summary of the Disease and Its Treatments Paperback; Independently Published: San Diego, CA, USA, 2019; ISBN 978-1-79586-060-4.
- Wasif, N.; Maggard, M.A.; Ko, C.Y.; Giuliano, A.E. Invasive Lobular vs. Ductal Breast Cancer: A Stage-Matched Comparison of Outcomes. Ann. Surg. Oncol. 2010, 17, 1862–1869. [CrossRef]
- 9. Holland, D.W.; Boucher, L.D.; Mortimer, J.E. Tubular Breast Cancer Experience at Washington University: A Review of the Literature. *Clin. Breast Cancer* 2001, 2, 210–214. [CrossRef]
- 10. Anwar, S.L.; Dwianingsih, E.K.; Avanti, W.S.; Choridah, L.; Suwardjo; Aryandono, T. Aggressive Behavior of Her-2 Positive Colloid Breast Carcinoma: A Case Report in a Metastatic Breast Cancer. *Ann. Med. Surg.* **2020**, *52*, 48–52. [CrossRef]
- 11. Singletary, S.E.; Connolly, J.L. Breast Cancer Staging: Working With the Sixth Edition of the AJCC Cancer Staging Manual. *CA Cancer J. Clin.* **2006**, *56*, 37–47. [CrossRef]
- 12. Veale, D.; Ashcroft, T.; Marsh, C.; Gibson, G.; Harris, A. Epidermal Growth Factor Receptors in Non-Small Cell Lung Cancer. *Br. J. Cancer* **1987**, *55*, 513–516. [CrossRef] [PubMed]
- Singletary, S.E.; Allred, C.; Ashley, P.; Bassett, L.W.; Berry, D.; Bland, K.I.; Borgen, P.I.; Clark, G.; Edge, S.B.; Hayes, D.F.; et al. Revision of the American Joint Committee on Cancer Staging System for Breast Cancer. J. Clin. Oncol. 2002, 20, 3628–3636. [CrossRef]
- 14. Wang, R.; Zhu, Y.; Liu, X.; Liao, X.; He, J.; Niu, L. The Clinicopathological Features and Survival Outcomes of Patients with Different Metastatic Sites in Stage IV Breast Cancer. *BMC Cancer* **2019**, *19*, 1091. [CrossRef]
- 15. Sung, H.; Ferlay, J.; Siegel, R.L.; Laversanne, M.; Soerjomataram, I.; Jemal, A.; Bray, F. Global Cancer Statistics 2020: GLOBOCAN Estimates of Incidence and Mortality Worldwide for 36 Cancers in 185 Countries. *CA Cancer J. Clin.* **2021**, *71*, 209–249. [CrossRef]
- 16. Finkel, M.L. Breast Cancer Facts, Myths, and Controversies: Understanding Current Screenings and Treatments; Public Health Issues and Developments; ABC-CLIO: Santa Barbara, CA, USA, 2021; ISBN 978-1-4408-7513-7.
- 17. Miller, K.D.; Camp, M.; Steligo, K. *The Breast Cancer Book: A Trusted Guide for You and Your Loved Ones*; A Johns Hopkins Press Health Book; Johns Hopkins University Press: Baltimore, MA, USA, 2021; ISBN 978-1-4214-4190-0.
- Link, J.S.; Ein-Gal, S.; Link, N. The Breast Cancer Survival Manual: A Step-by-Step Guide for Women with Newly Diagnosed Breast Cancer, 6th ed.; Completely revised and updated, First St. Martin's Griffin edition; St. Martin's Griffin: New York, NY, USA, 2017; ISBN 978-1-250-14452-2.
- 19. Lee, K.W.; Chung, S.Y.; Yang, I.; Kim, H.D.; Shin, S.J.; Kim, J.E.; Chung, B.W.; Choi, J.-A. Inflammatory Breast Cancer. *Clin. Imaging* 2005, *29*, 22–25. [CrossRef] [PubMed]
- Chen, D.; Zhao, H.; He, J.; Pan, Q.; Zhao, W. An Causal XAI Diagnostic Model for Breast Cancer Based on Mammography Reports. In Proceedings of the 2021 IEEE International Conference on Bioinformatics and Biomedicine (BIBM), Houston, TX, USA, 9 December 2021; IEEE: Piscataway, NJ, USA, 2021; pp. 3341–3349.
- 21. Elmore, J.G. Screening for Breast Cancer. JAMA 2005, 293, 1245. [CrossRef] [PubMed]
- 22. Kunkel, E.J.; Chen, E.I.; Okunlola, T.B. Psychosocial Concerns of Women with Breast Cancer. *Prim. Care Update OBGYNS* 2002, *9*, 129–134. [CrossRef]
- 23. O'Malley, M.S. Screening for Breast Cancer With Breast Self-Examination: A Critical Review. JAMA 1987, 257, 2196. [CrossRef]
- Akram, M.; Iqbal, M.; Daniyal, M.; Khan, A.U. Awareness and Current Knowledge of Breast Cancer. *Biol. Res.* 2017, 50, 33. [CrossRef]
- Shahari, S.; Wakankar, A. Color Analysis of Thermograms for Breast Cancer Detection. In Proceedings of the 2015 International Conference on Industrial Instrumentation and Control (ICIC), Pune, India, 28–30 May 2015; IEEE: Piscataway, NJ, USA, 2015; pp. 1577–1581.
- Rizza, P.; Pellegrino, M.; Caruso, A.; Iacopetta, D.; Sinicropi, M.S.; Rault, S.; Lancelot, J.C.; El-Kashef, H.; Lesnard, A.; Rochais, C.; et al. 3-(Dipropylamino)-5-Hydroxybenzofuro[2,3-f] Quinazolin-1(2H)-One (DPA-HBFQ-1) Plays an Inhibitory Role on Breast Cancer Cell Growth and Progression. *Eur. J. Med. Chem.* 2016, 107, 275–287. [CrossRef]
- 27. Chu, C.-N.; Hu, K.-C.; Wu, R.S.-C.; Bau, D.-T. Radiation-Irritated Skin and Hyperpigmentation May Impact the Quality of Life of Breast Cancer Patients after Whole Breast Radiotherapy. *BMC Cancer* **2021**, *21*, 330. [CrossRef] [PubMed]
- Roointan, A.; Ahmad Mir, T.; Ibrahim Wani, S.; Mati-ur-Rehman; Hussain, K.K.; Ahmed, B.; Abrahim, S.; Savardashtaki, A.; Gandomani, G.; Gandomani, M.; et al. Early Detection of Lung Cancer Biomarkers through Biosensor Technology: A Review. J. Pharm. Biomed. Anal. 2019, 164, 93–103. [CrossRef] [PubMed]
- Kothari, R.; Fong, Y.; Storrie-Lombardi, M.C. Review of Laser Raman Spectroscopy for Surgical Breast Cancer Detection: Stochastic Backpropagation Neural Networks. *Sensors* 2020, 20, 6260. [CrossRef] [PubMed]
- Taha, S.; Mohamed, W.R.; Elhemely, M.A.; El-Gendy, A.O.; Mohamed, T. Tunable Femtosecond Laser Suppresses the Proliferation of Breast Cancer in Vitro. J. Photochem. Photobiol. B 2023, 240, 112665. [CrossRef]
- Zeeshan, M.; Salam, B.; Khalid, Q.S.B.; Alam, S.; Sayani, R. Diagnostic Accuracy of Digital Mammography in the Detection of Breast Cancer. *Cureus* 2018, 10, e2448. [CrossRef]
- 32. Jaglan, P.; Dass, R.; Duhan, M. Breast Cancer Detection Techniques: Issues and Challenges. J. Inst. Eng. India Ser. B 2019, 100, 379–386. [CrossRef]
- Sood, R.; Rositch, A.F.; Shakoor, D.; Ambinder, E.; Pool, K.-L.; Pollack, E.; Mollura, D.J.; Mullen, L.A.; Harvey, S.C. Ultrasound for Breast Cancer Detection Globally: A Systematic Review and Meta-Analysis. J. Glob. Oncol. 2019, 2019, 1–17. [CrossRef]
- 34. Yang, S.K.; Cho, N.; Moon, W.K. The Role of PET/CT for Evaluating Breast Cancer. Korean J. Radiol. 2007, 8, 429–437. [CrossRef]

- Avril, N.; Rosé, C.A.; Schelling, M.; Dose, J.; Kuhn, W.; Bense, S.; Weber, W.; Ziegler, S.; Graeff, H.; Schwaiger, M. Breast Imaging With Positron Emission Tomography and Fluorine-18 Fluorodeoxyglucose: Use and Limitations. *J. Clin. Oncol.* 2000, 18, 3495–3502. [CrossRef]
- Desperito, E.; Schwartz, L.; Capaccione, K.M.; Collins, B.T.; Jamabawalikar, S.; Peng, B.; Patrizio, R.; Salvatore, M.M. Chest CT for Breast Cancer Diagnosis. *Life* 2022, 12, 1699. [CrossRef]
- 37. Bickelhaupt, S.; Steudle, F.; Paech, D.; Mlynarska, A.; Kuder, T.A.; Lederer, W.; Daniel, H.; Freitag, M.; Delorme, S.; Schlemmer, H.-P.; et al. On a Fractional Order Calculus Model in Diffusion Weighted Breast Imaging to Differentiate between Malignant and Benign Breast Lesions Detected on X-ray Screening Mammography. *PLoS ONE* 2017, *12*, e0176077. [CrossRef] [PubMed]
- Zhao, Y.; Brun, E.; Coan, P.; Huang, Z.; Sztrókay, A.; Diemoz, P.C.; Liebhardt, S.; Mittone, A.; Gasilov, S.; Miao, J.; et al. High-Resolution, Low-Dose Phase Contrast X-ray Tomography for 3D Diagnosis of Human Breast Cancers. *Proc. Natl. Acad. Sci. USA* 2012, 109, 18290–18294. [CrossRef] [PubMed]
- Wishart, G.C.; Campisi, M.; Boswell, M.; Chapman, D.; Shackleton, V.; Iddles, S.; Hallett, A.; Britton, P.D. The Accuracy of Digital Infrared Imaging for Breast Cancer Detection in Women Undergoing Breast Biopsy. *Eur. J. Surg. Oncol. EJSO* 2010, *36*, 535–540. [CrossRef]
- Hartman, J.; Ström, A.; Gustafsson, J.-Å. Estrogen Receptor Beta in Breast Cancer—Diagnostic and Therapeutic Implications. Steroids 2009, 74, 635–641. [CrossRef] [PubMed]
- Yip, C.-H.; Rhodes, A. Estrogen and Progesterone Receptors in Breast Cancer. *Future Oncol.* 2014, 10, 2293–2301. [CrossRef] [PubMed]
- Lambertini, M.; Kroman, N.; Ameye, L.; Cordoba, O.; Pinto, A.; Benedetti, G.; Jensen, M.-B.; Gelber, S.; Del Grande, M.; Ignatiadis, M.; et al. Long-Term Safety of Pregnancy Following Breast Cancer According to Estrogen Receptor Status. *JNCI J. Natl. Cancer Inst.* 2018, 110, 426–429. [CrossRef]
- 43. Sauter, G.; Lee, J.; Bartlett, J.M.S.; Slamon, D.J.; Press, M.F. Guidelines for Human Epidermal Growth Factor Receptor 2 Testing: Biologic and Methodologic Considerations. *J. Clin. Oncol.* **2009**, *27*, 1323–1333. [CrossRef]
- 44. Cleator, S.; Heller, W.; Coombes, R.C. Triple-Negative Breast Cancer: Therapeutic Options. *Lancet Oncol.* 2007, *8*, 235–244. [CrossRef]
- 45. Wang, L. Microwave Sensors for Breast Cancer Detection. Sensors 2018, 18, 655. [CrossRef]
- Akbari, A.; Hashemzadeh, H.; Eshkiki, Z.S.; Masoodi, M.; Tabaeian, S.P.; Naderi-Manesh, H.; Zare, A.A.; Agah, S. Detection of Plasma MiR-223 by a Novel Label-Free Graphene Oxide/Gold Nanocomposite Immunosensor in Colorectal Cancer Patients: An Electrochemical Biosensor Approach. *Biosens. Bioelectron. X* 2023, 14, 100331. [CrossRef]
- Hashemzadeh, H.; Khadivi-Khanghah, Z.; Allahverdi, A.; Hadipour, M.M.; Saievar-Iranizad, E.; Naderi-Manesh, H. A Novel Label-Free Graphene Oxide Nano-Wall Surface Decorated with Gold Nano-Flower Biosensor for Electrochemical Detection of Brucellosis Antibodies in Human Serum. *Talanta Open* 2023, 7, 100215. [CrossRef]
- Parchekani, J.; Hashemzadeh, H.; Allahverdi, A.; Siampour, H.; Abbasian, S.; Moshaii, A.; Naderi-Manesh, H. Zepto Molar MiRNA-21 Detection in Gold Nano-Islands Platform toward Early Cancer Screening. *Sens. Bio-Sens. Res.* 2021, 34, 100449. [CrossRef]
- Ross, J.S.; Linette, G.P.; Stec, J.; Clark, E.; Ayers, M.; Leschly, N.; Symmans, W.F.; Hortobagyi, G.N.; Pusztai, L. Breast Cancer Biomarkers and Molecular Medicine. *Expert Rev. Mol. Diagn.* 2003, *3*, 573–585. [CrossRef] [PubMed]
- Nicolini, A.; Ferrari, P.; Duffy, M.J. Prognostic and Predictive Biomarkers in Breast Cancer: Past, Present and Future. Semin. Cancer Biol. 2018, 52, 56–73. [CrossRef] [PubMed]
- 51. Adhami, M.; Haghdoost, A.A.; Sadeghi, B.; Malekpour Afshar, R. Candidate MiRNAs in Human Breast Cancer Biomarkers: A Systematic Review. *Breast Cancer* 2018, 25, 198–205. [CrossRef] [PubMed]
- 52. Hong, R.; Sun, H.; Li, D.; Yang, W.; Fan, K.; Liu, C.; Dong, L.; Wang, G. A Review of Biosensors for Detecting Tumor Markers in Breast Cancer. *Life* 2022, *12*, 342. [CrossRef]
- Fliss, A.E.; Benzeno, S.; Rao, J.; Caplan, A.J. Control of Estrogen Receptor Ligand Binding by Hsp90. J. Steroid Biochem. Mol. Biol. 2000, 72, 223–230. [CrossRef]
- 54. Grøntved, L.; Hager, G.L. Impact of Chromatin Structure on PR Signaling: Transition from Local to Global Analysis. *Mol. Cell. Endocrinol.* **2012**, 357, 30–36. [CrossRef]
- Iqbal, N.; Iqbal, N. Human Epidermal Growth Factor Receptor 2 (HER2) in Cancers: Overexpression and Therapeutic Implications. *Mol. Biol. Int.* 2014, 2014, 1–9. [CrossRef]
- Cardinali, B.; Tasso, R.; Piccioli, P.; Ciferri, M.C.; Quarto, R.; Del Mastro, L. Circulating MiRNAs in Breast Cancer Diagnosis and Prognosis. *Cancers* 2022, 14, 2317. [CrossRef]
- Heller, S.L.; Moy, L. MRI Breast Screening Revisited: MRI Breast Screening Revisited. J. Magn. Reson. Imaging 2019, 49, 1212–1221. [CrossRef] [PubMed]
- Saslow, D.; Boetes, C.; Burke, W.; Harms, S.; Leach, M.O.; Lehman, C.D.; Morris, E.; Pisano, E.; Schnall, M.; Sener, S.; et al. American Cancer Society Guidelines for Breast Screening with MRI as an Adjunct to Mammography. *CA Cancer J. Clin.* 2007, 57, 75–89. [CrossRef] [PubMed]
- Jain, M.; Jain, A.; Hyzy, M.D.; Werth, G. FAST MRI Breast Screening Revisited. J. Med. Imaging Radiat. Oncol. 2017, 61, 24–28. [CrossRef] [PubMed]

- 60. Menezes, G.L. Magnetic Resonance Imaging in Breast Cancer: A Literature Review and Future Perspectives. *World J. Clin. Oncol.* **2014**, *5*, 61. [CrossRef] [PubMed]
- Peters, N.H.G.M.; Borel Rinkes, I.H.M.; Zuithoff, N.P.A.; Mali, W.P.T.M.; Moons, K.G.M.; Peeters, P.H.M. Meta-Analysis of MR Imaging in the Diagnosis of Breast Lesions. *Radiology* 2008, 246, 116–124. [CrossRef]
- Turnbull, L.W. Dynamic Contrast-Enhanced MRI in the Diagnosis and Management of Breast Cancer. NMR Biomed. 2009, 22, 28–39. [CrossRef]
- 63. Mahmud, M.; Islam, M.; Misran, N.; Almutairi, A.; Cho, M. Ultra-Wideband (UWB) Antenna Sensor Based Microwave Breast Imaging: A Review. *Sensors* 2018, *18*, 2951. [CrossRef]
- 64. Qashlan, A.M.; Aldhaheri, R.W.; Alharbi, K.H. A Modified Compact Flexible Vivaldi Antenna Array Design for Microwave Breast Cancer Detection. *Appl. Sci.* 2022, 12, 4908. [CrossRef]
- 65. Elsheakh, D.N.; Mohamed, R.A.; Fahmy, O.M.; Ezzat, K.; Eldamak, A.R. Complete Breast Cancer Detection and Monitoring System by Using Microwave Textile Based Antenna Sensors. *Biosensors* **2023**, *13*, 87. [CrossRef]
- Amdaouch, I.; Saban, M.; El Gueri, J.; Chaari, M.Z.; Alejos, A.V.; Alzola, J.R.; Muñoz, A.R.; Aghzout, O. A Novel Approach of a Low-Cost UWB Microwave Imaging System with High Resolution Based on SAR and a New Fast Reconstruction Algorithm for Early-Stage Breast Cancer Detection. *J. Imaging* 2022, *8*, 264. [CrossRef]
- Lim, S.; Yoon, Y.J. Wideband-Narrowband Switchable Tapered Slot Antenna for Breast Cancer Diagnosis and Treatment. *Appl. Sci.* 2021, 11, 3606. [CrossRef]
- 68. *Electromagnetic Waves and Antennas for Biomedical Applications;* Wang, L. (Ed.) Healthcare technologies series; Institution of Engineering and Technology: Stevenage, UK, 2022; ISBN 978-1-83953-167-5.
- 69. Alamro, W.; Seet, B.-C.; Wang, L.; Parthiban, P. Early-Stage Lung Tumor Detection Based on Super-Wideband Microwave Reflectometry. *Electronics* **2022**, *12*, *36*. [CrossRef]
- 70. Wang, L. Multi-Frequency Holographic Microwave Imaging for Breast Lesion Detection. *IEEE Access* **2019**, *7*, 83984–83993. [CrossRef]
- Lazebnik, M.; McCartney, L.; Popovic, D.; Watkins, C.B.; Lindstrom, M.J.; Harter, J.; Sewall, S.; Magliocco, A.; Booske, J.H.; Okoniewski, M.; et al. A Large-Scale Study of the Ultrawideband Microwave Dielectric Properties of Normal Breast Tissue Obtained from Reduction Surgeries. *Phys. Med. Biol.* 2007, *52*, 2637–2656. [CrossRef]
- 72. Wang, L. Terahertz Imaging for Breast Cancer Detection. Sensors 2021, 21, 6465. [CrossRef]
- Wang, L.; Xu, J. A New Wideband Microwave Antenna for Breast Cancer Detection. In Proceedings of the Volume 3: Biomedical and Biotechnology Engineering, Pittsburgh, PA, USA, 9–15 November 2018; p. V003T04A040.
- Lazebnik, M.; Popovic, D.; McCartney, L.; Watkins, C.B.; Lindstrom, M.J.; Harter, J.; Sewall, S.; Ogilvie, T.; Magliocco, A.; Breslin, T.M.; et al. A Large-Scale Study of the Ultrawideband Microwave Dielectric Properties of Normal, Benign and Malignant Breast Tissues Obtained from Cancer Surgeries. *Phys. Med. Biol.* 2007, 52, 6093–6115. [CrossRef]
- O'Loughlin, D.; Elahi, M.A.; Porter, E.; Shahzad, A.; Oliveira, B.L.; Glavin, M.; Jones, E.; O'Halloran, M. Open-Source Software for Microwave Radar-Based Image Reconstruction. In Proceedings of the 12th European Conference on Antennas and Propagation (EuCAP 2018), London, UK, 9–13 April 2018; p. 408.
- Kerhet, A.; Raffetto, M.; Boni, A.; Massa, A. A SVM-Based Approach to Microwave Breast Cancer Detection. *Eng. Appl. Artif. Intell.* 2006, 19, 807–818. [CrossRef]
- Sechopoulos, I.; Mann, R.M. Stand-Alone Artificial Intelligence—The Future of Breast Cancer Screening? *Breast* 2020, 49, 254–260. [CrossRef]
- Abdul Halim, A.A.; Andrew, A.M.; Mohd Yasin, M.N.; Abd Rahman, M.A.; Jusoh, M.; Veeraperumal, V.; Rahim, H.A.; Illahi, U.; Abdul Karim, M.K.; Scavino, E. Existing and Emerging Breast Cancer Detection Technologies and Its Challenges: A Review. *Appl. Sci.* 2021, *11*, 10753. [CrossRef]
- Woten, D.A.; Lusth, J.; El-Shenawee, M. Interpreting Artificial Neural Networks for Microwave Detection of Breast Cancer. IEEE Microw. Wirel. Compon. Lett. 2007, 17, 825–827. [CrossRef]
- Syed, A.; Sobahi, N.; Sheikh, M.; Mittra, R.; Rmili, H. Modified 16-Quasi Log Periodic Antenna Array for Microwave Imaging of Breast Cancer Detection. *Appl. Sci.* 2021, 12, 147. [CrossRef]
- 81. Subramanian, S.; Sundarambal, B.; Nirmal, D. Investigation on Simulation-Based Specific Absorption Rate in Ultra-Wideband Antenna for Breast Cancer Detection. *IEEE Sens. J.* 2018, *18*, 10002–10009. [CrossRef]
- 82. Moussa, M.N.; Madi, M.A.; Kabalan, K.Y. Breast Tumor Detection, Sizing and Localization Using a 24-Element Antenna Array. *IEEE J. Biomed. Health Inform.* **2022**, *26*, 5109–5121. [CrossRef] [PubMed]
- Alsharif, F.; Kurnaz, C. Wearable Microstrip Patch Ultra Wide Band Antenna for Breast Cancer Detection. In Proceedings of the 2018 41st International Conference on Telecommunications and Signal Processing (TSP), Athens, Greece, 4–6 July 2018; IEEE: Piscataway, NJ, USA, 2018; pp. 1–5.
- Soltani, M.; Rahpeima, R.; Kashkooli, F.M. Breast Cancer Diagnosis with a Microwave Thermoacoustic Imaging Technique—A Numerical Approach. *Med. Biol. Eng. Comput.* 2019, 57, 1497–1513. [CrossRef]
- Rahpeima, R.; Soltani, M.; Moradi Kashkooli, F. Numerical Study of Microwave Induced Thermoacoustic Imaging for Initial Detection of Cancer of Breast on Anatomically Realistic Breast Phantom. *Comput. Methods Programs Biomed.* 2020, 196, 105606. [CrossRef]

- 86. Yan, A.; Lin, L.; Liu, C.; Shi, J.; Na, S.; Wang, L.V. Microwave-induced Thermoacoustic Tomography through an Adult Human Skull. *Med. Phys.* **2019**, *46*, 1793–1797. [CrossRef]
- 87. Cui, Y.; Yuan, C.; Ji, Z. A Review of Microwave-Induced Thermoacoustic Imaging: Excitation Source, Data Acquisition System and Biomedical Applications. J. Innov. Opt. Health Sci. 2017, 10, 1730007. [CrossRef]
- Lou, C.; Yang, S.; Ji, Z.; Chen, Q.; Xing, D. Ultrashort Microwave-Induced Thermoacoustic Imaging: A Breakthrough in Excitation Efficiency and Spatial Resolution. *Phys. Rev. Lett.* 2012, 109, 218101. [CrossRef]
- 89. Wang, B.; Sun, Y.; Wang, Z.; Wang, X. Three-Dimensional Microwave-Induced Thermoacoustic Imaging Based on Compressive Sensing Using an Analytically Constructed Dictionary. *IEEE Trans. Microw. Theory Tech.* **2020**, *68*, 377–386. [CrossRef]
- Eletxigerra, U.; Martinez-Perdiguero, J.; Barderas, R.; Pingarrón, J.M.; Campuzano, S.; Merino, S. Surface Plasmon Resonance Immunosensor for ErbB2 Breast Cancer Biomarker Determination in Human Serum and Raw Cancer Cell Lysates. *Anal. Chim. Acta* 2016, 905, 156–162. [CrossRef]
- 91. Szymanska, B.; Lukaszewski, Z.; Hermanowicz-Szamatowicz, K.; Gorodkiewicz, E. An Immunosensor for the Determination of Carcinoembryonic Antigen by Surface Plasmon Resonance Imaging. *Anal. Biochem.* **2020**, *609*, 113964. [CrossRef] [PubMed]
- Ghosh, S.K.; Pal, T. Interparticle Coupling Effect on the Surface Plasmon Resonance of Gold Nanoparticles: From Theory to Applications. *Chem. Rev.* 2007, 107, 4797–4862. [CrossRef]
- 93. Daghestani, H.N.; Day, B.W. Theory and Applications of Surface Plasmon Resonance, Resonant Mirror, Resonant Waveguide Grating, and Dual Polarization Interferometry Biosensors. *Sensors* **2010**, *10*, 9630–9646. [CrossRef] [PubMed]
- Kamal Eddin, F.B.; Fen, Y.W. The Principle of Nanomaterials Based Surface Plasmon Resonance Biosensors and Its Potential for Dopamine Detection. *Molecules* 2020, 25, 2769. [CrossRef]
- 95. Abdelghaffar, M.; Gamal, Y.; El-Khoribi, R.A.; Soliman, W.; Badr, Y.; Hameed, M.F.O.; Obayya, S.S.A. Highly Sensitive V-Shaped SPR PCF Biosensor for Cancer Detection. *Opt. Quantum Electron.* **2023**, *55*, 472. [CrossRef]
- 96. Yasli, A. Cancer Detection with Surface Plasmon Resonance-Based Photonic Crystal Fiber Biosensor. *Plasmonics* **2021**, *16*, 1605–1612. [CrossRef]
- Park, S.-C.; Cho, E.-J.; Moon, S.-Y.; Yoon, S.-I.; Kim, Y.-J.; Kim, D.-H.; Suh, J.-S. A Calorimetric Biosensor and Its Application for Detecting a Cancer Cell with Optical Imaging. In *World Congress on Medical Physics and Biomedical Engineering* 2006; Magjarevic, R., Nagel, J.H., Eds.; IFMBE Proceedings; Springer: Berlin/Heidelberg, Germany, 2007; Volume 14, pp. 637–640. ISBN 978-3-540-36839-7.
- 98. Bai, Y.; Li, H.; Xu, J.; Huang, Y.; Zhang, X.; Weng, J.; Li, Z.; Sun, L. Ultrasensitive Colorimetric Biosensor for BRCA1 Mutation Based on Multiple Signal Amplification Strategy. *Biosens. Bioelectron.* **2020**, *166*, 112424. [CrossRef]
- 99. Choi, J.-H.; Lim, J.; Shin, M.; Paek, S.-H.; Choi, J.-W. CRISPR-Cas12a-Based Nucleic Acid Amplification-Free DNA Biosensor via Au Nanoparticle-Assisted Metal-Enhanced Fluorescence and Colorimetric Analysis. *Nano Lett.* **2021**, *21*, 693–699. [CrossRef]
- Manikandan, M.; Nasser Abdelhamid, H.; Talib, A.; Wu, H.-F. Facile Synthesis of Gold Nanohexagons on Graphene Templates in Raman Spectroscopy for Biosensing Cancer and Cancer Stem Cells. *Biosens. Bioelectron.* 2014, 55, 180–186. [CrossRef]
- Ding, S.-Y.; You, E.-M.; Tian, Z.-Q.; Moskovits, M. Electromagnetic Theories of Surface-Enhanced Raman Spectroscopy. *Chem. Soc. Rev.* 2017, 46, 4042–4076. [CrossRef]
- Chen, K.-H.; Pan, M.-J.; Jargalsaikhan, Z.; Ishdorj, T.-O.; Tseng, F.-G. Development of Surface-Enhanced Raman Scattering (SERS)-Based Surface-Corrugated Nanopillars for Biomolecular Detection of Colorectal Cancer. *Biosensors* 2020, 10, 163. [CrossRef] [PubMed]
- 103. Téllez-Plancarte, A.; Haro-Poniatowski, E.; Picquart, M.; Morales-Méndez, J.; Lara-Cruz, C.; Jiménez-Salazar, J.; Damián-Matsumura, P.; Escobar-Alarcón, L.; Batina, N. Development of a Nanostructured Platform for Identifying HER2-Heterogeneity of Breast Cancer Cells by Surface-Enhanced Raman Scattering. *Nanomaterials* 2018, *8*, 549. [CrossRef] [PubMed]
- Guo, L.; Fu, F.; Chen, G. Capillary Electrophoresis with Electrochemiluminescence Detection: Fundamental Theory, Apparatus, and Applications. Anal. Bioanal. Chem. 2011, 399, 3323–3343. [CrossRef] [PubMed]
- 105. Hu, L.; Xu, G. Applications and Trends in Electrochemiluminescence. Chem. Soc. Rev. 2010, 39, 3275. [CrossRef]
- 106. Liu, Y.; Zhang, H.; Li, B.; Liu, J.; Jiang, D.; Liu, B.; Sojic, N. Single Biomolecule Imaging by Electrochemiluminescence. J. Am. Chem. Soc. 2021, 143, 17910–17914. [CrossRef]
- 107. Lu, Y.; Rong, X.; Wu, P.; Shou, J.; Chen, L.; Luo, F.; Lin, C.; Wang, J.; Qiu, B.; Lin, Z. Sensitive Electrochemiluminescence Biosensor Based on the Target Trigger Difference of the Electrostatic Interaction between an ECL Reporter and the Electrode Surface. *Anal. Chem.* 2022, 94, 5823–5829. [CrossRef] [PubMed]
- Nasrollahpour, H.; Mahdipour, M.; Isildak, I.; Rashidi, M.-R.; Naseri, A.; Khalilzadeh, B. A Highly Sensitive Electrochemiluminescence Cytosensor for Detection of SKBR-3 Cells as Metastatic Breast Cancer Cell Line: A Constructive Phase in Early and Precise Diagnosis. *Biosens. Bioelectron.* 2021, 178, 113023. [CrossRef]
- Xie, J.; Mu, Z.; Qing, M.; Zhou, J.; Sun, S.; Bai, L. A Novel Binary Luminophore Based High-Efficient Electrochemiluminescence Biosensor for Ultrasensitive Detection of Human Epidermal Growth Factor Receptor-2. *Chem. Eng. J.* 2022, 450, 138362. [CrossRef]
- 110. Nakamoto, T.; Moriizumi, T. A Theory of a Quartz Crystal Microbalance Based upon a Mason Equivalent Circuit. *Jpn. J. Appl. Phys.* **1990**, *29*, 963. [CrossRef]
- Yang, X.; Zhou, R.; Hao, Y.; Yang, P. A CD44-Biosensor for Evaluating Metastatic Potential of Breast Cancer Cells Based on Quartz Crystal Microbalance. Sci. Bull. 2017, 62, 923–930. [CrossRef]

- 112. Wang, X.; Yu, H.; Lu, D.; Zhang, J.; Deng, W. Label Free Detection of the Breast Cancer Biomarker CA15.3 Using ZnO Nanorods Coated Quartz Crystal Microbalance. *Sens. Actuators B Chem.* **2014**, *195*, 630–634. [CrossRef]
- Lin, S.; Wang, J.; Lin, Y.; Wang, X. Biosensitivity of Molybdenum Disulfide for Monitoring Breast Cancer Marker CA15-3 Using Quartz Crystal Microbalance. Int. J. Electrochem. Sci. 2021, 16, 150712. [CrossRef]
- Wang, Y.; Wei, Z.; Luo, X.; Wan, Q.; Qiu, R.; Wang, S. An Ultrasensitive Homogeneous Aptasensor for Carcinoembryonic Antigen Based on Upconversion Fluorescence Resonance Energy Transfer. *Talanta* 2019, 195, 33–39. [CrossRef] [PubMed]
- 115. Jeong, Y.; Kook, Y.-M.; Lee, K.; Koh, W.-G. Metal Enhanced Fluorescence (MEF) for Biosensors: General Approaches and a Review of Recent Developments. *Biosens. Bioelectron.* **2018**, *111*, 102–116. [CrossRef]
- 116. Mohammadi, S.; Mohammadi, S.; Salimi, A. A 3D Hydrogel Based on Chitosan and Carbon Dots for Sensitive Fluorescence Detection of MicroRNA-21 in Breast Cancer Cells. *Talanta* **2021**, 224, 121895. [CrossRef]
- Sadeghi, S.; Rahaie, M. Design and Fabrication of a DNA-Copper Nanocluster-Based Biosensor for Multiple Detections of Circulating MiRNAs in Early Screening of Breast Cancer. J. Fluoresc. 2022, 32, 2297–2307. [CrossRef]
- 118. Henel, G.; Schmitz, J. Basic Theory and Clinical Applications of Flow Cytometry. Lab. Med. 2007, 38, 428–436. [CrossRef]
- 119. Shapiro, H.M. Practical Flow Cytometry, 4th ed.; Wiley-Liss: New York, NY, USA, 2003; ISBN 978-0-471-41125-3.
- 120. Givan, A.L. Flow Cytometry: First Principles, 2nd ed.; Wiley-Liss: New York, NY, USA, 2001; ISBN 978-0-471-38224-9.
- 121. Zand, E.; Froehling, A.; Schoenher, C.; Zunabovic-Pichler, M.; Schlueter, O.; Jaeger, H. Potential of Flow Cytometric Approaches for Rapid Microbial Detection and Characterization in the Food Industry—A Review. *Foods* **2021**, *10*, 3112. [CrossRef]
- 122. Cruz, I.; Ciudad, J.; Cruz, J.J.; Ramos, M.; Gómez-Alonso, A.; Adansa, J.C.; Rodríguez, C.; Orfao, A. Evaluation of Multiparameter Flow Cytometry for the Detection of Breast Cancer Tumor Cells in Blood Samples. *Am. J. Clin. Pathol.* **2005**, *123*, 66–74. [CrossRef]
- 123. Lostumbo, A.; Mehta, D.; Setty, S.; Nunez, R. Flow Cytometry: A New Approach for the Molecular Profiling of Breast Cancer. *Exp. Mol. Pathol.* **2006**, *80*, 46–53. [CrossRef]
- 124. Bruus, H. *Theoretical Microfluidics*; Oxford Master Series in Physics; Oxford University Press: Oxford, UK; New York, NY, USA, 2008; ISBN 978-0-19-923508-7.
- Gwak, H.; Park, S.; Kim, J.; Lee, J.D.; Kim, I.-S.; Kim, S.-I.; Hyun, K.-A.; Jung, H.-I. Microfluidic Chip for Rapid and Selective Isolation of Tumor-Derived Extracellular Vesicles for Early Diagnosis and Metastatic Risk Evaluation of Breast Cancer. *Biosens. Bioelectron.* 2021, 192, 113495. [CrossRef] [PubMed]
- 126. Gao, H.; Yan, C.; Wu, W.; Li, J. Application of Microfluidic Chip Technology in Food Safety Sensing. *Sensors* 2020, 20, 1792. [CrossRef] [PubMed]
- Chen, W.; Li, H.; Su, W.; Qin, J. Microfluidic Device for On-Chip Isolation and Detection of Circulating Exosomes in Blood of Breast Cancer Patients. *Biomicrofluidics* 2019, 13, 054113. [CrossRef] [PubMed]
- 128. Fang, S.; Tian, H.; Li, X.; Jin, D.; Li, X.; Kong, J.; Yang, C.; Yang, X.; Lu, Y.; Luo, Y.; et al. Clinical Application of a Microfluidic Chip for Immunocapture and Quantification of Circulating Exosomes to Assist Breast Cancer Diagnosis and Molecular Classification. *PLoS ONE* 2017, 12, e0175050. [CrossRef] [PubMed]
- 129. Gao, Y.; Qiang, L.; Chu, Y.; Han, Y.; Zhang, Y.; Han, L. Microfluidic Chip for Multiple Detection of MiRNA Biomarkers in Breast Cancer Based on Three-Segment Hybridization. *AIP Adv.* **2020**, *10*, 045022. [CrossRef]
- Pacheco, J.G.; Rebelo, P.; Freitas, M.; Nouws, H.P.A.; Delerue-Matos, C. Breast Cancer Biomarker (HER2-ECD) Detection Using a Molecularly Imprinted Electrochemical Sensor. Sens. Actuators B Chem. 2018, 273, 1008–1014. [CrossRef]
- You, M.; Yang, S.; Tang, W.; Zhang, F.; He, P. Molecularly Imprinted Polymers-Based Electrochemical DNA Biosensor for the Determination of BRCA-1 Amplified by SiO2@Ag. *Biosens. Bioelectron.* 2018, 112, 72–78. [CrossRef]
- 132. Saylan, Y.; Akgönüllü, S.; Yavuz, H.; Ünal, S.; Denizli, A. Molecularly Imprinted Polymer Based Sensors for Medical Applications. *Sensors* 2019, 19, 1279. [CrossRef]
- 133. Humairah, N.A.; Nurijal, I.; Ainus Sofa, S.; Chaerunnisa, A.; Roto, R.; Kusumaatmaja, A.; Hadi Sujiono, E.; Triyana, K. Molecularly Imprinted Polyvinyl Acetate Doped with Boric Acid for Sensitivity and Selectivity of Ammonia Sensing by QCM. *Biosens. Bioelectron. X* 2023, 13, 100320. [CrossRef]
- Ribeiro, J.A.; Pereira, C.M.; Silva, A.F.; Sales, M.G.F. Disposable Electrochemical Detection of Breast Cancer Tumour Marker CA 15-3 Using Poly (Toluidine Blue) as Imprinted Polymer Receptor. *Biosens. Bioelectron.* 2018, 109, 246–254. [CrossRef]
- 135. Li, T.; Liang, Y.; Li, J.; Yu, Y.; Xiao, M.-M.; Ni, W.; Zhang, Z.; Zhang, G.-J. Carbon Nanotube Field-Effect Transistor Biosensor for Ultrasensitive and Label-Free Detection of Breast Cancer Exosomal MiRNA21. Anal. Chem. 2021, 93, 15501–15507. [CrossRef] [PubMed]
- Shao, N.; Wickstrom, E.; Panchapakesan, B. Nanotube–Antibody Biosensor Arrays for the Detection of Circulating Breast Cancer Cells. Nanotechnology 2008, 19, 465101. [CrossRef] [PubMed]
- Novodchuk, I.; Bajcsy, M.; Yavuz, M. Graphene-Based Field Effect Transistor Biosensors for Breast Cancer Detection: A Review on Biosensing Strategies. *Carbon* 2021, 172, 431–453. [CrossRef]
- Reiner-Rozman, C.; Larisika, M.; Nowak, C.; Knoll, W. Graphene-Based Liquid-Gated Field Effect Transistor for Biosensing: Theory and Experiments. *Biosens. Bioelectron.* 2015, 70, 21–27. [CrossRef]
- 139. Thriveni, G.; Ghosh, K. Advancement and Challenges of Biosensing Using Field Effect Transistors. *Biosensors* 2022, *12*, 647. [CrossRef]

- 140. Tran, D.P.; Winter, M.A.; Wolfrum, B.; Stockmann, R.; Yang, C.-T.; Pourhassan-Moghaddam, M.; Offenhäusser, A.; Thierry, B. Toward Intraoperative Detection of Disseminated Tumor Cells in Lymph Nodes with Silicon Nanowire Field Effect Transistors. ACS Nano 2016, 10, 2357–2364. [CrossRef]
- Manimekala, T.; Sivasubramanian, R.; Dharmalingam, G. Nanomaterial-Based Biosensors Using Field-Effect Transistors: A Review. J. Electron. Mater. 2022, 51, 1950–1973. [CrossRef]
- 142. Shivaram, K.B.; Bhatt, P.; Verma, M.S.; Clase, K.; Simsek, H. Bacteriophage-Based Biosensors for Detection of Pathogenic Microbes in Wastewater. *Sci. Total Environ.* 2023, 901, 165859. [CrossRef]
- 143. Han, A.; Yang, L.; Frazier, A.B. Quantification of the Heterogeneity in Breast Cancer Cell Lines Using Whole-Cell Impedance Spectroscopy. *Clin. Cancer Res.* 2007, 13, 139–143. [CrossRef]
- 144. Vadhva, P.; Hu, J.; Johnson, M.J.; Stocker, R.; Braglia, M.; Brett, D.J.L.; Rettie, A.J.E. Electrochemical Impedance Spectroscopy for All-Solid-State Batteries: Theory, Methods and Future Outlook. *ChemElectroChem* 2021, *8*, 1930–1947. [CrossRef]
- 145. Ciucci, F. Modeling Electrochemical Impedance Spectroscopy. Curr. Opin. Electrochem. 2019, 13, 132–139. [CrossRef]
- 146. Lakhera, P.; Chaudhary, V.; Jha, A.; Singh, R.; Kush, P.; Kumar, P. Recent Developments and Fabrication of the Different Electrochemical Biosensors Based on Modified Screen Printed and Glassy Carbon Electrodes for the Early Diagnosis of Diverse Breast Cancer Biomarkers. *Mater. Today Chem.* 2022, 26, 101129. [CrossRef]
- Mollarasouli, F.; Kurbanoglu, S.; Ozkan, S.A. The Role of Electrochemical Immunosensors in Clinical Analysis. *Biosensors* 2019, 9, 86. [CrossRef] [PubMed]
- 148. Huerta-Nuñez, L.F.E.; Gutierrez-Iglesias, G.; Martinez-Cuazitl, A.; Mata-Miranda, M.M.; Alvarez-Jiménez, V.D.; Sánchez-Monroy, V.; Golberg, A.; González-Díaz, C.A. A Biosensor Capable of Identifying Low Quantities of Breast Cancer Cells by Electrical Impedance Spectroscopy. *Sci. Rep.* 2019, *9*, 6419. [CrossRef] [PubMed]
- Marques, R.C.B.; Viswanathan, S.; Nouws, H.P.A.; Delerue-Matos, C.; González-García, M.B. Electrochemical Immunosensor for the Analysis of the Breast Cancer Biomarker HER2 ECD. *Talanta* 2014, 129, 594–599. [CrossRef] [PubMed]
- Wang, W.; Lim, L.G.; Srivastava, S.; Yan, J.S.B.; Shabbir, A.; Liu, Q. Roles of Linear and Circular Polarization Properties and Effect of Wavelength Choice on Differentiation between *Ex Vivo* Normal and Cancerous Gastric Samples. *J. Biomed. Opt.* 2014, 19, 046020. [CrossRef]
- 151. Tromberg, B.J.; Coquoz, O.; Fishkin, J.B.; Pham, T.; Anderson, E.R.; Butler, J.; Cahn, M.; Gross, J.D.; Venugopalan, V.; Pham, D. Non–Invasive Measurements of Breast Tissue Optical Properties Using Frequency–Domain Photon Migration. *Philos. Trans. R. Soc. Lond. B Biol. Sci.* **1997**, 352, 661–668. [CrossRef]
- 152. No, K.S.; Chou, P.H. Mini-FDPM and Heterodyne Mini-FDPM: Handheld Non-Invasive Breast Cancer Detectors Based on Frequency-Domain Photon Migration. *IEEE Trans. Circuits Syst. Regul. Pap.* **2005**, *52*, 2672–2685. [CrossRef]
- 153. Carp, S.A.; Kauffman, T.; Fang, Q.; Rafferty, E.; Moore, R.; Kopans, D.; Boas, D. Compression-Induced Changes in the Physiological State of the Breast as Observed through Frequency Domain Photon Migration Measurements. J. Biomed. Opt. 2006, 11, 064016. [CrossRef]
- 154. No, K.S.; Xie, Q.; Kwong, R.; Cerussi, A.; Tromberg, B.J.; Chou, P.H. HBS: A Handheld Breast Cancer Detector Based on Frequency Domain Photon Migration with Full Heterodyne. In Proceedings of the 2006 IEEE Biomedical Circuits and Systems Conference, London, UK, 29 November–1 December 2006; IEEE: Piscataway, NJ, USA, 2006; pp. 114–117.
- 155. Bowman, T.C.; El-Shenawee, M.; Campbell, L.K. Terahertz Imaging of Excised Breast Tumor Tissue on Paraffin Sections. *IEEE Trans. Antennas Propag.* 2015, 63, 2088–2097. [CrossRef]
- Bowman, T.; Wu, Y.; Gauch, J.; Campbell, L.K.; El-Shenawee, M. Terahertz Imaging of Three-Dimensional Dehydrated Breast Cancer Tumors. J. Infrared Millim. Terahertz Waves 2017, 38, 766–786. [CrossRef]
- 157. Li, X.; Ma, F.; Yang, M.; Zhang, J. Nanomaterial Based Analytical Methods for Breast Cancer Biomarker Detection. *Mater. Today Adv.* **2022**, *14*, 100219. [CrossRef]
- 158. Xu, J.; Zhao, C.; Niu, K.; Gao, Z.; Song, Y.-Y. Renewable Photoelectrochemical Cytosensing Platform for Rapid Capture and Detection of Circulating Tumor Cells. *Anal. Chim. Acta* 2021, 1142, 1–9. [CrossRef] [PubMed]
- 159. Zhou, Y.; Zheng, Y.; Zhu, X.; Chai, Y.; Yuan, R. Modular Engineering of Gold-Silver Nanocluster Supermolecular Structure Endow Strong Electrochemiluminescence for Ultrasensitive Bioanalysis. *Biosens. Bioelectron.* **2021**, *190*, 113449. [CrossRef] [PubMed]
- Wei, X.; Duan, X.; Zhou, X.; Wu, J.; Xu, H.; Min, X.; Ding, S. A Highly Sensitive SPRi Biosensing Strategy for Simultaneous Detection of Multiplex MiRNAs Based on Strand Displacement Amplification and AuNP Signal Enhancement. *Analyst* 2018, 143, 3134–3140. [CrossRef] [PubMed]
- Jia, S.; Li, P.; Koh, K.; Chen, H. A Cytosensor Based on NiO Nanoparticle-Enhanced Surface Plasmon Resonance for Detection of the Breast Cancer Cell Line MCF-7. *Microchim. Acta* 2016, 183, 683–688. [CrossRef]
- Song, X.; Xu, T.; Song, Y.; He, X.; Wang, D.; Liu, C.; Zhang, X. Droplet Array for Open-Channel High-Throughput SERS Biosensing. *Talanta* 2020, 218, 121206. [CrossRef]
- Wang, X.; Cheng, S.; Wang, X.; Wei, L.; Kong, Q.; Ye, M.; Luo, X.; Xu, J.; Zhang, C.; Xian, Y. PH-Sensitive Dye-Based Nanobioplatform for Colorimetric Detection of Heterogeneous Circulating Tumor Cells. ACS Sens. 2021, 6, 1925–1932. [CrossRef]
- Ahirwar, R.; Nahar, P. Development of a Label-Free Gold Nanoparticle-Based Colorimetric Aptasensor for Detection of Human Estrogen Receptor Alpha. *Anal. Bioanal. Chem.* 2016, 408, 327–332. [CrossRef]

- 165. Zhang, Y.; Jiao, J.; Wei, Y.; Wang, D.; Yang, C.; Xu, Z. Plasmonic Colorimetric Biosensor for Sensitive Exosome Detection via Enzyme-Induced Etching of Gold Nanobipyramid@MnO₂ Nanosheet Nanostructures. *Anal. Chem.* 2020, 92, 15244–15252. [CrossRef]
- 166. Nassif, A.B.; Talib, M.A.; Nasir, Q.; Afadar, Y.; Elgendy, O. Breast Cancer Detection Using Artificial Intelligence Techniques: A Systematic Literature Review. *Artif. Intell. Med.* **2022**, 127, 102276. [CrossRef] [PubMed]
- Freeman, K.; Geppert, J.; Stinton, C.; Todkill, D.; Johnson, S.; Clarke, A.; Taylor-Phillips, S. Use of Artificial Intelligence for Image Analysis in Breast Cancer Screening Programmes: Systematic Review of Test Accuracy. *BMJ* 2021, 374, n1872. [CrossRef] [PubMed]
- 168. Houssami, N.; Kirkpatrick-Jones, G.; Noguchi, N.; Lee, C.I. Artificial Intelligence (AI) for the Early Detection of Breast Cancer: A Scoping Review to Assess AI's Potential in Breast Screening Practice. *Expert Rev. Med. Devices* 2019, 16, 351–362. [CrossRef] [PubMed]
- Shah, S.M.; Khan, R.A.; Arif, S.; Sajid, U. Artificial Intelligence for Breast Cancer Analysis: Trends & Directions. Comput. Biol. Med. 2022, 142, 105221. [CrossRef] [PubMed]
- 170. Kim, H.-E.; Kim, H.H.; Han, B.-K.; Kim, K.H.; Han, K.; Nam, H.; Lee, E.H.; Kim, E.-K. Changes in Cancer Detection and False-Positive Recall in Mammography Using Artificial Intelligence: A Retrospective, Multireader Study. *Lancet Digit. Health* 2020, 2, e138–e148. [CrossRef] [PubMed]
- 171. Carter, S.M.; Rogers, W.; Win, K.T.; Frazer, H.; Richards, B.; Houssami, N. The Ethical, Legal and Social Implications of Using Artificial Intelligence Systems in Breast Cancer Care. *Breast* 2020, *49*, 25–32. [CrossRef]
- 172. Wang, L. Deep Learning Techniques to Diagnose Lung Cancer. Cancers 2022, 14, 5569. [CrossRef]
- 173. Wang, L. Holographic Microwave Image Classification Using a Convolutional Neural Network. *Micromachines* **2022**, *13*, 2049. [CrossRef]
- 174. Wang, L.; Xu, J. Deep Convolutional Neural Networks for Breast Image Analysis on Holographic Microwave Imaging. In Proceedings of the Volume 3: Biomedical and Biotechnology Engineering, Pittsburgh, PA, USA, 9–15 November 2018; p. V003T04A029.
- 175. Kangas, M.J.; Burks, R.M.; Atwater, J.; Lukowicz, R.M.; Williams, P.; Holmes, A.E. Colorimetric Sensor Arrays for the Detection and Identification of Chemical Weapons and Explosives. *Crit. Rev. Anal. Chem.* **2017**, *47*, 138–153. [CrossRef]
- 176. Yoo, S.-M.; Jeon, Y.-M.; Heo, S.-Y. Electrochemiluminescence Systems for the Detection of Biomarkers: Strategical and Technological Advances. *Biosensors* 2022, *12*, 738. [CrossRef]
- 177. Heydari, S.; Haghayegh, G.H. Application of Nanoparticles in Quartz Crystal Microbalance Biosensors. J. Sens. Technol. 2014, 04, 81–100. [CrossRef]
- 178. Kampik, M.; Klonz, M.; Skubis, T. Thermal Converter with Quartz Crystal Temperature Sensor for Ac-Dc Transfer. *IEEE Trans. Instrum. Meas.* **1997**, *46*, 387–390. [CrossRef]
- 179. Zhang, X.; Xiao, Y.; Qian, X. A Ratiometric Fluorescent Probe Based on FRET for Imaging Hg²⁺ Ions in Living Cells. *Angew. Chem. Int. Ed.* **2008**, 47, 8025–8029. [CrossRef] [PubMed]
- Yin, J.; Li, C.; Wang, D.; Liu, S. FRET-Derived Ratiometric Fluorescent K ⁺ Sensors Fabricated from Thermoresponsive Poly(*N*-Isopropylacrylamide) Microgels Labeled with Crown Ether Moieties. *J. Phys. Chem. B* 2010, *114*, 12213–12220. [CrossRef] [PubMed]
- 181. Hussain, S.A.; Dey, D.; Chakraborty, S.; Saha, J.; Roy, A.D.; Chakraborty, S.; Debnath, P.; Bhattacharjee, D. Fluorescence Resonance Energy Transfer (FRET) Sensor. *arXiv* 2014, arXiv:1408.6559. [CrossRef]
- Park, J.; Nguyen, H.H.; Woubit, A.; Kim, M. Applications of Field-Effect Transistor (FET)-Type Biosensors. *Appl. Sci. Converg. Technol.* 2014, 23, 61–71. [CrossRef]
- 183. Loyez, M.; Lobry, M.; Hassan, E.M.; DeRosa, M.C.; Caucheteur, C.; Wattiez, R. HER2 Breast Cancer Biomarker Detection Using a Sandwich Optical Fiber Assay. *Talanta* **2021**, *221*, 121452. [CrossRef]
- Chen, H.; Hou, Y.; Ye, Z.; Wang, H.; Koh, K.; Shen, Z.; Shu, Y. Label-Free Surface Plasmon Resonance Cytosensor for Breast Cancer Cell Detection Based on Nano-Conjugation of Monodisperse Magnetic Nanoparticle and Folic Acid. Sens. Actuators B Chem. 2014, 201, 433–438. [CrossRef]
- Zheng, Z.; Wu, L.; Li, L.; Zong, S.; Wang, Z.; Cui, Y. Simultaneous and Highly Sensitive Detection of Multiple Breast Cancer Biomarkers in Real Samples Using a SERS Microfluidic Chip. *Talanta* 2018, 188, 507–515. [CrossRef]
- 186. Guerrero-Esteban, T.; Gutiérrez-Sánchez, C.; García-Mendiola, T.; Revenga-Parra, M.; Pariente, F.; Lorenzo, E. Bifunctional Carbon Nanodots for Highly Sensitive HER2 Determination Based on Electrochemiluminescence. *Sens. Actuators B Chem.* 2021, 343, 130096. [CrossRef]
- Motaghi, H.; Ziyaee, S.; Mehrgardi, M.A.; Kajani, A.A.; Bordbar, A.-K. Electrochemiluminescence Detection of Human Breast Cancer Cells Using Aptamer Modified Bipolar Electrode Mounted into 3D Printed Microchannel. *Biosens. Bioelectron.* 2018, 118, 217–223. [CrossRef] [PubMed]
- Zhang, X.; Ke, H.; Wang, Z.; Guo, W.; Zhang, A.; Huang, C.; Jia, N. An Ultrasensitive Multi-Walled Carbon Nanotube–Platinum–Luminol Nanocomposite-Based Electrochemiluminescence Immunosensor. *Analyst* 2017, 142, 2253–2260. [CrossRef] [PubMed]
- Liao, H.; Zhou, Y.; Chai, Y.; Yuan, R. An Ultrasensitive Electrochemiluminescence Biosensor for Detection of MicroRNA by In-Situ Electrochemically Generated Copper Nanoclusters as Luminophore and TiO2 as Coreaction Accelerator. *Biosens. Bioelectron.* 2018, 114, 10–14. [CrossRef] [PubMed]

- Zhang, A.; Xiang, H.; Zhang, X.; Guo, W.; Yuan, E.; Huang, C.; Jia, N. A Novel Sandwich Electrochemiluminescence Immunosensor for Ultrasensitive Detection of Carbohydrate Antigen 19-9 Based on Immobilizing Luminol on Ag@BSA Core/Shell Microspheres. *Biosens. Bioelectron.* 2016, 75, 206–212. [CrossRef] [PubMed]
- Jiang, X.; Wang, H.; Wang, H.; Zhuo, Y.; Yuan, R.; Chai, Y. Electrochemiluminescence Biosensor Based on 3-D DNA Nanomachine Signal Probe Powered by Protein-Aptamer Binding Complex for Ultrasensitive Mucin 1 Detection. *Anal. Chem.* 2017, *89*, 4280–4286. [CrossRef]
- 192. Nasrollahpour, H.; Isildak, I.; Rashidi, M.-R.; Hashemi, E.A.; Naseri, A.; Khalilzadeh, B. Ultrasensitive Bioassaying of HER-2 Protein for Diagnosis of Breast Cancer Using Reduced Graphene Oxide/Chitosan as Nanobiocompatible Platform. *Cancer* Nanotechnol. 2021, 12, 10. [CrossRef]
- Zhang, S.; Bai, H.; Luo, J.; Yang, P.; Cai, J. A Recyclable Chitosan-Based QCM Biosensor for Sensitive and Selective Detection of Breast Cancer Cells in Real Time. *The Analyst* 2014, 139, 6259–6265. [CrossRef]
- 194. Chang, K.; Pi, Y.; Lu, W.; Wang, F.; Pan, F.; Li, F.; Jia, S.; Shi, J.; Deng, S.; Chen, M. Label-Free and High-Sensitive Detection of Human Breast Cancer Cells by Aptamer-Based Leaky Surface Acoustic Wave Biosensor Array. *Biosens. Bioelectron.* 2014, 60, 318–324. [CrossRef]
- Bakhshpour, M.; Piskin, A.K.; Yavuz, H.; Denizli, A. Quartz Crystal Microbalance Biosensor for Label-Free MDA MB 231 Cancer Cell Detection via Notch-4 Receptor. *Talanta* 2019, 204, 840–845. [CrossRef]
- Attia, M.S.; Ali, K.; El-Kemary, M.; Darwish, W.M. Phthalocyanine-Doped Polystyrene Fluorescent Nanocomposite as a Highly Selective Biosensor for Quantitative Determination of Cancer Antigen 125. *Talanta* 2019, 201, 185–193. [CrossRef]
- 197. Ji, H.; Zhang, X.; Dai, Y.; Xue, T.; Misal, S.; Qi, Z. A Highly Selective Ratiometric Fluorescent Probe Based on Naphthalimide for Detection and Imaging of CYP1A1 in Living Cells and Zebrafish. *Analyst* 2019, 144, 7390–7397. [CrossRef] [PubMed]
- 198. Li, Q.; Zhou, S.; Zhang, T.; Zheng, B.; Tang, H. Bioinspired Sensor Chip for Detection of MiRNA-21 Based on Photonic Crystals Assisted Cyclic Enzymatic Amplification Method. *Biosens. Bioelectron.* **2020**, *150*, 111866. [CrossRef] [PubMed]
- Li, Y.; Tang, D.; Zhu, L.; Cai, J.; Chu, C.; Wang, J.; Xia, M.; Cao, Z.; Zhu, H. Label-Free Detection of MiRNA Cancer Markers Based on Terminal Deoxynucleotidyl Transferase-Induced Copper Nanoclusters. *Anal. Biochem.* 2019, 585, 113346. [CrossRef] [PubMed]
- Arya, S.K.; Zhurauski, P.; Jolly, P.; Batistuti, M.R.; Mulato, M.; Estrela, P. Capacitive Aptasensor Based on Interdigitated Electrode for Breast Cancer Detection in Undiluted Human Serum. *Biosens. Bioelectron.* 2018, 102, 106–112. [CrossRef]
- De Oliveira, R.A.G.; Materon, E.M.; Melendez, M.E.; Carvalho, A.L.; Faria, R.C. Disposable Microfluidic Immunoarray Device for Sensitive Breast Cancer Biomarker Detection. ACS Appl. Mater. Interfaces 2017, 9, 27433–27440. [CrossRef] [PubMed]
- 202. Fragoso, A.; Latta, D.; Laboria, N.; Von Germar, F.; Hansen-Hagge, T.E.; Kemmner, W.; Gärtner, C.; Klemm, R.; Drese, K.S.; O'Sullivan, C.K. Integrated Microfluidic Platform for the Electrochemical Detection of Breast Cancer Markers in Patient Serum Samples. Lab Chip 2011, 11, 625–631. [CrossRef]
- Liu, C.; Xu, X.; Li, B.; Situ, B.; Pan, W.; Hu, Y.; An, T.; Yao, S.; Zheng, L. Single-Exosome-Counting Immunoassays for Cancer Diagnostics. *Nano Lett.* 2018, 18, 4226–4232. [CrossRef]
- Li, L.; Wang, Q.; Feng, J.; Tong, L.; Tang, B. Highly Sensitive and Homogeneous Detection of Membrane Protein on a Single Living Cell by Aptamer and Nicking Enzyme Assisted Signal Amplification Based on Microfluidic Droplets. *Anal. Chem.* 2014, *86*, 5101–5107. [CrossRef]
- 205. Lahcen, A.A.; Rauf, S.; Aljedaibi, A.; De Oliveira Filho, J.I.; Beduk, T.; Mani, V.; Alshareef, H.N.; Salama, K.N. Laser-Scribed Graphene Sensor Based on Gold Nanostructures and Molecularly Imprinted Polymers: Application for Her-2 Cancer Biomarker Detection. *Sens. Actuators B Chem.* 2021, 347, 130556. [CrossRef]
- Pacheco, J.G.; Silva, M.S.V.; Freitas, M.; Nouws, H.P.A.; Delerue-Matos, C. Molecularly Imprinted Electrochemical Sensor for the Point-of-Care Detection of a Breast Cancer Biomarker (CA 15-3). Sens. Actuators B Chem. 2018, 256, 905–912. [CrossRef]
- 207. Majd, S.M.; Salimi, A.; Ghasemi, F. Ultrasensitive Detection of MiRNA-155 in Breast Cancer via Direct Hybridization Assay Using Two-Dimensional Molybdenum Disulfide Field-Effect Transistor Biosensor. *Biosens. Bioelectron.* 2018, 105, 6–13. [CrossRef] [PubMed]
- Myung, S.; Solanki, A.; Kim, C.; Park, J.; Kim, K.S.; Lee, K.-B. Graphene-Encapsulated Nanoparticle-Based Biosensor for the Selective Detection of Cancer Biomarkers. *Adv. Mater.* 2011, 23, 2221–2225. [CrossRef] [PubMed]
- Reyes, P.I.; Ku, C.-J.; Duan, Z.; Lu, Y.; Solanki, A.; Lee, K.-B. ZnO Thin Film Transistor Immunosensor with High Sensitivity and Selectivity. *Appl. Phys. Lett.* 2011, 98, 173702. [CrossRef]
- Mansouri Majd, S.; Salimi, A. Ultrasensitive Flexible FET-Type Aptasensor for CA 125 Cancer Marker Detection Based on Carboxylated Multiwalled Carbon Nanotubes Immobilized onto Reduced Graphene Oxide Film. *Anal. Chim. Acta* 2018, 1000, 273–282. [CrossRef] [PubMed]
- Wang, W.; Fan, X.; Xu, S.; Davis, J.J.; Luo, X. Low Fouling Label-Free DNA Sensor Based on Polyethylene Glycols Decorated with Gold Nanoparticles for the Detection of Breast Cancer Biomarkers. *Biosens. Bioelectron.* 2015, 71, 51–56. [CrossRef]
- 212. Ferreira, D.C.; Batistuti, M.R.; Bachour, B.; Mulato, M. Aptasensor Based on Screen-Printed Electrode for Breast Cancer Detection in Undiluted Human Serum. *Bioelectrochemistry* **2021**, *137*, 107586. [CrossRef] [PubMed]
- Cardoso, A.R.; Moreira, F.T.C.; Fernandes, R.; Sales, M.G.F. Novel and Simple Electrochemical Biosensor Monitoring Attomolar Levels of MiRNA-155 in Breast Cancer. *Biosens. Bioelectron.* 2016, 80, 621–630. [CrossRef]

- Zhang, K.; Zhang, N.; Zhang, L.; Wang, H.; Shi, H.; Liu, Q. Label-Free Impedimetric Sensing Platform for MicroRNA-21 Based on ZrO₂-Reduced Graphene Oxide Nanohybrids Coupled with Catalytic Hairpin Assembly Amplification. *RSC Adv.* 2018, *8*, 16146–16151. [CrossRef]
- Elshafey, R.; Tavares, A.C.; Siaj, M.; Zourob, M. Electrochemical Impedance Immunosensor Based on Gold Nanoparticles–Protein G for the Detection of Cancer Marker Epidermal Growth Factor Receptor in Human Plasma and Brain Tissue. *Biosens. Bioelectron.* 2013, 50, 143–149. [CrossRef]
- 216. Ravalli, A.; Da Rocha, C.G.; Yamanaka, H.; Marrazza, G. A Label-Free Electrochemical Affisensor for Cancer Marker Detection: The Case of HER2. *Bioelectrochemistry* **2015**, *106*, 268–275. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.